

## **Comparison of modelling results across scales**

## Deliverable D3.2

April 2018

Lisa Nowak<sup>1</sup>, Ylva Longva<sup>1</sup>, Mark Rounsevell<sup>1</sup>, Paula Harrison<sup>2</sup>, Pei-Yuan Chen<sup>2</sup>, Robert Dunford<sup>2</sup>, Ian Holman<sup>3</sup>, Lamprini Papadimitriou<sup>3</sup>, Stefan Fronzek<sup>4</sup>, Tim Carter<sup>4</sup>, Emma Terämä<sup>4</sup>, Calum Brown<sup>1</sup>, Harald Bugmann<sup>5</sup>, Rebecca Snell<sup>6</sup>, Valentine Lafond<sup>5</sup>, Magnus Benzie<sup>7</sup>, Anastasia Lobanova<sup>8</sup>, Valentina Krysanova<sup>8</sup>, Florian Wimmer<sup>9</sup>, Sen Li<sup>10</sup>

> <sup>1</sup>University of Edinburgh, UK <sup>2</sup>Centre for Ecology & Hydrology, UK <sup>3</sup>Cranfield University, UK <sup>4</sup>SYKE, Finland <sup>5</sup>ETH-Zurich, Switzerland <sup>6</sup>University of Ohio, USA <sup>7</sup>SEI, Sweden <sup>8</sup>PIK, Germany <sup>9</sup>Universität Kassel, Germany <sup>10</sup>University of Oxford, UK

**IMPRESSIONS** – Impacts and Risks from High-End Scenarios: Strategies for Innovative Solutions (www.impressions-project.eu)



#### Prepared under contract from the European Commission

Contract n° 603416 Collaborative project FP7 Environment

Project acronym: Project full title: Start of the project: Duration:	IMPRESSIONS Impacts and Risks from High-end Scenarios: Strategies for Innovative Solutions 01 November 2013 60 months
Project coordinator:	NERC Centre for Ecology & Hydrology
Project website	www.impressions-project.eu
Deliverable title:	Comparison of modelling results across scales
Deliverable n°:	D3.2
Nature of the deliverable:	Report
Dissemination level:	Public
WP responsible:	WP3
Lead beneficiary:	UEDIN
Citation:	Nowak, L., Longva, Y., Rounsevell, M., Harrison, P., Chen, P., Dunford, R., Holman, I., Papadimitriou, L., Fronzek, S., Carter, T., Terämä, E., Brown, C., Bugmann, H., Snell, R., Lafond, V., Benzie, M., Lobanova, A., Krysanova, V., Wimmer, F. and Li, S. (2018). <i>Comparison of modelling results across scales</i> . EU FP7 IMPRESSIONS Project Deliverable D3.2
Due date of deliverable:	Month 50
Actual submission date:	Month 54

#### Deliverable status:

				_
Version	Status	Date	Author(s)	
1.0	Draft	28 March 2018	Nowak et al.	
2.0	Final draft	18 April 2018	Nowak et al.	
3.0	Final	20 April 2018	Nowak et al.	
-				

The content of this deliverable do not necessarily reflect the official opinions of the European Commission or other institutions of the European Union.

## **Table of contents**

Pre	face			. 5
Sun	nmary	·		. 6
1.	1. Introduction			
1	.1.	Desc	cription of work	. 9
1	.2.	Links	s to other work packages	. 9
2. Methods				. 9
2	.1.	Clim	ate and socio-economic scenarios	10
2	.2.	Desc	cription of case studies at different scales	11
	2.2.1	L.	Europe	11
	2.2.2	2.	Scotland	11
	2.2.3	3.	Iberia	11
	2.2.4	1.	Hungary	12
	2.2.5	5.	Central Asia (EU external): Implication for the EU of cross-border climate change	12
2	.3.	Euro	ppean sub-regions	12
3.	Qua	ntitat	tive inter-model and cross-scale comparisons	14
3	.1.	Crop	o yields	14
	3.1.1	L.	Inter-model comparison	14
	3.1.2	2.	Results	15
3	.2.	Fore	est productivity	18
	3.2.1	L.	Inter-model comparison	18
	3.2.2	2.	Results	19
3	.3.	Wat	er	23
	3.3.1	L.	Inter-model comparison	23
	3.3.2	2.	Results	24
	3.3.3	3.	Conclusions	26
3	.4.	Land	l use	26
	3.4.1	L.	Inter-model comparison	26
	3.4.2	2.	Results	27
3	.5.	Рорі	ulation and urban growth	33
	3.5.1	L.	Inter-model comparison	33
	3.5.2	2.	Results	35
4.	Inter	r-moo	del and cross-scale comparisons using impact response surfaces	43

5.	Qu	alitat	tive synthesis across case studies and models 49	
	5.1.	scription of synthesis table		
	5.2.	Lan	nd use (agriculture/urban)	
	5.3.	For	restry	
	5.4.	Wa	ater	
	5.5.	Soc	cial and wellbeing	
6.	The	e Cen	ntral Asia (EU external) case study63	
	6.1.	Bac	ckground63	
	6.2.	Dir	rect impacts of climate change in Central Asia63	
	6.3.	Ind	lirect impacts for the European Union66	
	6.3	8.1.	Energy	
	6.3	3.2.	Trade	
	6.3	3.3.	Conflict and security67	
	6.3	8.4.	Migration	
7. Discussion				
	7.1.	Cro	oss-scale Synthesis of Impacts	
7.2. Model uncertainty			odel uncertainty	
	7.3.	Lim	nits to adaptation	
8.	Сог	nclusi	ion	
Acknowledgements				
Re	eferen	ices		
A	opend	lix		

#### Preface

The EU-funded FP7 project IMPRESSIONS (Impacts and Risks from High-end Scenarios: Strategies for Innovative Solutions) is an ambitious study of the risks and consequences for Europe of a runaway greenhouse effect and the options available for averting its most adverse effects. Focusing on high-end projections of future climate change and operating in the context of alternative socio-economic development pathways for Europe, the project seeks to simulate future impacts on natural resources, land use and societal well-being in Europe during the 21st century. It attempts this using a suite of single-sector and integrated multi-sector models that simulate the dynamics of climate change impacts and adaptation using an iterative, time-dependent approach up to 2100. The options for adaptive management, including transformative change, are guided by stakeholder-led visions of a sustainable and equitable Europe by 2100.

This deliverable compares and analyses the modelling of impacts across scales as part of Task 3.4. Results from the European case study are compared quantitatively and qualitatively to results from the three regional case studies (Scotland, Iberia and Hungary). Results from the Central Asia (EU external) case study are presented independently, but interpreted in terms of indirect effects on Europe.

#### **Summary**

This deliverable presents the results of a cross-scale and inter-model comparison and assessment of the wide selection of model applications completed in the IMPRESSIONS project for the different case studies. The analysis is divided into (i) quantitative, inter-model comparisons by sector, (ii) inter-model and cross-scale comparisons using impact response surfaces, and (iii) qualitative synthesis of a range of model output indicators across case studies in tables. The cross-scale comparison is based on the pan-European results, compared to the three regional/local case studies in Scotland, Iberia and Hungary. A further analysis synthesises results from the EU external (EUx) case study of Central Asia using semi-quantitative indicators.

The integrated assessment approach, as outlined in Deliverable D3.1 (Carter et al., 2015), provides the framework for this study. It ensures the inputs/outputs to models were harmonised to the greatest possible extent making a model inter-comparison feasible as well as supporting the integration of models. In addition, the integrated climate and socio-economic scenarios that combine the Representative Concentration Pathways (RCPs) and the Shared Socio-economic Pathways (SSPs) (see Deliverable D2.4 – Kok et al., 2018) linked all of the model applications.

The quantitative inter-model and cross-scale comparisons contain results for the key sectors: agriculture (crop yields); forestry; water; rural land use; and urban land use (including population growth). For most sectors, results from the IMPRESSIONS integrated assessment platform (IAP2) were compared to the outputs from other relevant models. The inter-model comparisons were useful in identifying climate change impacts that agree across models and areas where further research is needed. In addition, using a variety of sectoral models, as well as integrated cross-sectoral models, enables a fuller picture to be created of cross-sectoral interdependencies and provides greater contextual understanding of feedbacks between sectors. The inter-model comparison has shown that impacts under high-end climate change are going to be severe, however, models differ in their estimation of magnitude and sometimes also direction of change. This applies to the land use models, where different modelling assumptions underlying the CRAFTY and the IAP2 models lead to different expected land use impacts. However, these differences, rather than invalidating results, show the need to question underlying model assumptions and continuously research and adjust models to improve outcomes.

The assessment of model sensitivity using Impact Response Surfaces for European sub-regions showed a large variation of sensitivities across individual indicators and regions for most indicators. Exceptions are two river discharge indicators and, to a lesser extent, net primary production and agricultural land use, whose regional values clustered around a small range of sensitivities both for temperature and precipitation. North-eastern Europe showed increases in yields of all crops and basal area of all tree species, whereas Central and Eastern Europe showed decreases in these indicators. In regions of southern Europe (Iberian Peninsula, France and Mediterranean) indicators of river discharge and stem basal area (except Holm oak) were projected to decrease, whereas crop yields increased in these regions, where it was assumed that irrigation would compensate for decreases in precipitation.

The qualitative synthesis of land use indicators (changes in urban area, intensive agriculture, extensive agriculture, pasture area), crop yields (wheat, barley, maize), forestry (managed and unmanaged forest area), water indicators (water availability, water exploitation index, change in people flooded, changes in discharge and flooding for select river systems) and wellbeing indicators (Lyme disease risk, ecosystem services supply/demand gap, heat mortality) showed consistencies in the direction of impacts across case studies for most indicators. However, the magnitude of impacts varies by region. Scotland generally experiences less change than the other regions, with the exception of Lyme disease risk, which increases much more strongly and crop yields, which show beneficial increases across all scenarios. In comparison, Iberia experiences more negative impacts, particularly in the water and

wellbeing sectors. Water availability in Iberia decreases for all climate change scenarios, although water availability averaged across the whole of Europe increases. In addition, heat mortality in the population older than 75 years is projected to increase the most in Iberia.

The work reported here has built on the findings of previous deliverables (Deliverables D3B.2 – Holman et al., 2017 and Deliverable D3C.2 – Clarke et al., 2017) by bringing together the model results from different case studies and presenting a cross-scale and inter-model comparison. This has highlighted important similarities as well as divergences in the direction and magnitude of a variety of climate change impacts. In addition, through inter-model comparisons, it was possible to analyse the assumptions that go into individual models as well as highlighting model uncertainties. This is important as greater knowledge about uncertainties can support stakeholders in making decisions about adaptation measures to address the future impacts of high-end climate change.

### 1. Introduction

The overall objective of WP3 is to advance and apply multi-scale integrated assessment methods and models to quantify and understand climate change impacts, adaptation and vulnerability (CCIAV) associated with high-end scenarios for key economic, social and environmental sectors and their cross-sectoral interactions.

This Deliverable (D3.2) focuses on the cross-scale and inter-model comparison of a range of CCIAV models. Understanding CCIAV across scales is vital, as climate change impacts are likely to vary across scale, not only in direction, but also in impact. This deliverable therefore compares and synthesises results from the individual IMPRESSIONS case studies (Europe, Hungary, Scotland, Iberia and Central Asia) to highlight commonalities as well as divergences between the cases. Understanding these variations can enable decision-makers to implement spatially and contextually specific adaptation measures that are suitable for a specific socio-economic context.

Inter-model comparisons help uncover model- as well as scenario-related uncertainties. Uncertainty is a factor that needs careful consideration in any modelling exercise, and this is particularly the case for CCIAV modelling, where uncertainties not only exist concerning the individual models used, but also in relation to climate change and socio-economic scenario uncertainties. The inter-model as well as the cross-scale comparisons can therefore help identify uncertainties related to these issues. In addition, by using a variety of models, individual model uncertainties can be mediated by the strengths of other models. A better understanding of uncertainties is important to enable effective decision-making despite gaps in our knowledge.

An overview of the sensitivity analyses that were undertaken for a range of CCIAV models using Impact Response Surfaces (IRS) for European sub-regions is also presented within this deliverable. For an IRS, a sensitivity analysis of a model to systematic changes in key climatic and socio-economic drivers is conducted and the resulting impact variable is plotted as a surface comprising contour lines of equal response. The IRS approach provides an opportunity to test model performance across a wide range of conditions, including those found at the high-end of projected changes that may lie outside the conventional application of many models. The outcomes can then assist in summarising and comparing model behaviour across sectors and regions.

This deliverable compares the climate change impacts that result from applying the integrated assessment approach outlined in Deliverable D3.1 (Carter et al. 2015) across the multiple scales and models of the IMPRESSIONS project. The deliverable is divided into the following parts:

- Section 2 provides background to the integrated scenarios that were used with the CCIAV models and the case study regions.
- Section 3 presents a quantitative inter-model and cross-scale comparison for key sectors (agriculture, water, forestry, land use and urban).
- Section 4 presents an IRS inter-model and cross-scale comparison.
- Section 5 presents a qualitative synthesis of results across case studies and models, summarising and comparing the magnitude and direction of change in key indicators. Synthesis tables are used to visualise projected impacts across the different scenarios and highlight differences between regions.
- Section 6 presents a synthesis of results from the Central Asia (EU External) case study across the key sectors of energy, trade, conflict and security, and migration, and provides an interpretation of the possible indirect implications of such impacts for Europe.

#### **1.1.** Description of work

This deliverable is produced in fulfilment of Task 3.4: "Analysis and comparison of the CCIAV results across scales", which the Description of Work sets out to:

[...] undertake a comparison of the results from the different methods and models at the multiple scale levels using spatio-temporal windows of model outputs that are common to each level. The global scale modelled results will be compared with the pan-European results for the European window for a limited set of common output variables and for a common set of scenarios. Likewise the European model results will be compared with results for the three regional/local case studies. The inter-comparison will be used to examine the effect of spatial resolution on model outcomes and on the ranges of uncertainty in the assessment models. For example, impact response surfaces offer one technique for investigating the strength and direction of impact responses within and across regions. The effects of scenario resolution will also be analysed. Model results will be compared statistically to evaluate the significance of the differences between the scale levels and to identify particular locations where the models either agree or disagree in direction or magnitude of change.

As the model applications were selected to match the key challenges in each case study in consultation with local stakeholders, some indicators differ across scales or between case studies, making comparison more challenging in some cases.

#### **1.2.** Links to other work packages

Work Package (WP) 3 consists of modelling work from all three IMPRESSIONS scale levels; global (WP3A), European (WP3B) and regional/local (WP3C). More detailed descriptions of model results are presented in Deliverables D3A.1 (Carter et al., 2016), D3B.2 (Holman et al., 2017) and D3C.2 (Clarke et al., 2017). This deliverable builds on the multi-scale modelling framework described in Deliverable D3.1 (Carter et al., 2015). The integrated modelling approach underpinning the majority of the results presented in this deliverable is founded on close linkages with other parts of the IMPRESSIONS project. The scenario development (WP2) and the future visions (WP4/5) have informed the model applications.

#### 2. Methods

The different case studies follow a common methodological framework. This includes case study specific scenarios, which hinder or support climate change adaptation and mitigation actions, a vision for the case study areas in 2100 and case study specific pathways to reach this vision. The global and European case studies provide boundary conditions for the local case studies. Stakeholder workshops played an important role in all case studies. The input of stakeholders was used to develop case study specific scenarios and visions, as well as to identify key climate change adaptation and mitigation actions and sequence them within adaptation, mitigation and transformation pathways. Details of the visions and pathways that were developed for each case study are reported in Deliverable D4.2 (Hölscher et al., 2017).

For the modelling work conducted in IMPRESSIONS, an integrated assessment approach was developed and followed. This was based on the following guiding principles:

- Integration across scales and across sectors supported by the development of data dictionaries to harmonise model inputs/outputs;
- Using the data dictionaries to establish information flows across modelling scales;
- Accommodating scenarios as a starting point (across temporal/spatial scales and sectors);
- Defining what is needed by the models from the scenarios (both qualitative/quantitative);

- Simulating synergies and trade-offs in impacts across sectors and scales;
- Focusing on adaptation (based on limits to adaptation);
- Taking account of time dependencies in impacts and adaptation;
- Simulating the adaptation, mitigation and transformative solutions within the pathways as much as feasible;
- Synthesising and communicating model outputs to stakeholders.

To achieve these ambitions, the integrated approach is implemented through the following steps that address key research and policy-related questions defined jointly by researchers and stakeholders for each case study:

- 1. A conceptual framework based on a revised Driver-Pressure-State-Impact-Response (DPSIR) framework;
- 2. A modelling framework that identifies the relationships between the different models used in each case study, as well as the relationships across geographic scales;
- 3. A set of modelling protocols that guide the various model applications within each case study.

A full description of the integrated assessment approach is presented in Deliverable D3.1 (Carter et al., 2015), and a table of the models used for all the analysis in the IMPRESSIONS project is provided in the Annex to this deliverable (Table A1).

The specific methods used for the analysis in this deliverable are laid out in each individual section (Sections 3-6). A background to the scenarios and case studies that have informed the modelling in IMPRESSIONS is given below.

#### 2.1. Climate and socio-economic scenarios

The selected scenarios were based on the RCP x SSP scenario framework and serve as the basis for the modelling runs at different scales (see Deliverable D2.1; Kok et al., 2015 for a detailed description of the selection of climate and socio-economic scenarios for the project). Representative Concentration Pathways (RCP) 4.5 and 8.5 were chosen as they cover moderate as well as high-end climate change scenarios and go beyond the 2°C threshold set by the Paris Agreement at a global level. An additional set of low-end climate change scenarios (RCP2.6) were added for the European case study after the initial scenario selection to respond to a request from the European Commission after the Paris Agreement in December 2015. Four Shared Socio-economic Pathways (SSPs) were chosen, covering a diverse range of plausible future socio-economic conditions: SSP1, SSP3, SSP4 and SSP5. The RCPs and SSPs were combined by linking the SSP narratives to the likely climate change outcomes resulting in four integrated scenarios that represent high and low mitigation and adaptation challenges. The five integrated scenarios were:

- SSP1 x RCP4.5
- SSP3 x RCP4.5
- SSP3 x RCP8.5
- SSP4 x RCP4.5
- SSP5 x RCP8.5

The scenarios were downscaled and further developed through a stakeholder-led process. This process focused on four of the five scenarios for pragmatic reasons (excluding SSP3 x RCP4.5). This resulted in context specific scenarios for each case study. See Deliverables D2.2 (Kok & Pedde, 2016) and D2.3 (Madsen et al., 2016) for full overviews of the socio-economic and climate scenarios for each case study, respectively, and D2.4 (Kok et al., 2018) for the cross-scale analysis of the integrated climate and socio-economic scenarios.

#### 2.2. Description of case studies at different scales

A short overview is provided of the aims of each case studies and the models that were selected for application within the case study. This provides the necessary background to understand the intermodel and cross-scale comparisons provided in the next section.

#### 2.2.1. Europe

The European case study was designed to advance and improve climate change impact and adaptation modelling across Europe. The case study area covers the area of EU member states as well as Norway and Switzerland, but excludes Cyprus. It employed a variety of CCIAV models to explore climate change impacts for land use, water, forestry and human wellbeing sectors with input from global and European scale datasets.

Models used:

- Integrated models: European Integrated Assessment Platform 2 (IAP2), regional Integrated Assessment Model (rIAM) (multiple land and water sectors);
- Physically-based model: SWIM (hydrology);
- Statistical model: HEET (Heat stress mortality);
- Agent-based model: CRAFTY (land use change).

For a full description, see Deliverables D3B.1 (Holman et al., 2015) and D3B.2 (Holman et al., 2017).

#### 2.2.2. Scotland

The Scottish case study analysed impacts of climate change at the national scale. It focused on risks posed by high-end scenarios to land and water resource sectors, including agriculture, forestry, tourism, health and river flows.

Models used:

- Integrated model: Scottish IAP2 (multiple land and water sectors);
- Physically-based model: SWIM (hydrology);
- Agent-based model: LYR model (Lyme disease);
- Physically-based model: ForClim (forest productivity).

For a full description, see Deliverable D3C.1 (Rounsevell et al., 2015).

#### 2.2.3. Iberia

The Iberian case study focused on the transboundary river basin of the Tagus River and explored issued of transboundary water usage in a dry area. Land uses along the river basin, such as agriculture, forestry and energy production are vulnerable to land use changes upstream and were also explored in the case study.

Models used:

- Physically-based model: LandClim (agroforestry productivity);
- Physically-based model: SWIM (hydrology, including water transfers).

For a full description, see Deliverable D3C.1 (Rounsevell et al., 2015).

#### 2.2.4. Hungary

The Hungarian case study analysed climate change impacts at the local scale in two Hungarian communities: Szekszárd and Veszprém. Three major topics were explored: urban and agricultural land use, water availability and human health. Impacts of high-end climate change on Hungary are likely to be among the strongest in Europe, and the case study offered the opportunity to help key local stakeholders identify ways of increasing resilience to climate change.

Models used:

- Statistical model: ALLOCATION (population and urban land use);
- Agent-based model: LYR (Lyme disease);
- Physically-based model: SWIM (hydrology);
- Statistical model: HEET (Heat stress mortality);
- Agent-based model: Aporia (agricultural land use).

For a full description, see Deliverable D3C.1 (Rounsevell et al., 2015).

#### 2.2.5. Central Asia (EU external): Implication for the EU of cross-border climate change

This case study based in Central Asia explored high-end scenarios and cross-border impacts. This is an important field to explore as cross-border impact analysis is a relatively immature field of research, and the inclusion of a case study outside the borders of Europe allowed this to be developed. The case study focused on cross-border dynamics of the Central Asian republics of Kazakhstan, Uzbekistan, Turkmenistan, Kyrgyzstan and Tajikistan as well as taking into consideration geo-political dynamics with Russia and China (see Deliverable D3A.2; Benzie et al., 2017). The impacts of activities in Central Asia on the EU are analysed, but also what sort of adaptation strategies might be required to deal with external challenges. A more detailed account of the Central Asia case study is given in Section 5 of this deliverable.

#### 2.3. European sub-regions

Europe was sub-divided into eight sub-regions following Rockel & Worth (2007) for the synthesis and comparison of climate change impacts across scales (Figure 1). The same sub-regions were used for the sensitivity analysis conducted in the project using an impact response surface (IRS) approach (see Annex B of Deliverable D3.1 [Carter et al., 2015]; Fronzek et al., 2018; and Section 4 of this deliverable).

For this deliverable, the sub-regions of particular importance are:

- British Isles, used in some analyses as a proxy for the Scotland case study;
- Iberia, used in some analysis as a proxy for the Iberian case study;
- East Europe, used in some analysis as a proxy for the Hungarian case study.



Figure 1: European sub-regions. Source: Fronzek et al. (2018), modified from Rockel and Worth (2007).

#### 3. Quantitative inter-model and cross-scale comparisons

This section describes a set of detailed quantitative inter-model and cross-scale comparisons for the key sectors of: agriculture (crop yields); forestry; water; land use; and urban land use and population growth. Brief model descriptions are provided for each sector-based comparison within the following sub-sections.

#### 3.1. Crop yields

#### 3.1.1. Inter-model comparison

Crop yield data for Europe from the global models within the Inter-Sectoral Impact Model Intercomparison project (ISIMIP) were compared to outputs from the IMPRESSIONS IAP2. ISIMIP yield data for four crops (wheat, maize, soy and rapeseed) from seven crop models (EPIC, GEPIC, IMAGE, LPJ-GUESS, LPJmL, pDSSAT and PEGASUS) were extracted from those 0.5° x 0.5° grid cells within the IMPRESSIONS European domain (number of grid-cells varying between 3646 and 4979 across the different crop models). The IAP2 output has a finer spatial resolution of 10' x 10' (and 23871 grid-cells over Europe). A short introduction to the two model sets and results from the model intercomparison are given below.

#### 3.1.1.1 IMPRESSIONS Integrated Assessment Platform 2 (IAP2)

The IAP2 is an interactive, exploratory, web-based tool for simulating climate change impacts and vulnerabilities on a range of sectors (Harrison et al., 2018). The Platform integrates a suite of models of urban development, water resources, coasts, agriculture and forests, and biodiversity to simulate the spatial effects of different climatic and socio-economic scenarios across Europe. IAP2 builds on IAP1 (Harrison et al., 2015a), developed in the EU CLIMSAVE project, by integrating new climate and socio-economic scenarios and extending model projections to 2100. IAP1 has been applied widely in climate change impact, adaptation and vulnerability assessments, in robust policy analysis and has been tested extensively through model sensitivity and uncertainty analyses (Harrison et al., 2015; Dunford et al., 2015; Brown et al., 2015; Jäger et al., 2015; Kebede et al., 2015). The Platform operates at a spatial resolution of 10 arcmin x 10 arcmin grid-cells (and 23871 grid-cells over Europe), although multiple soil types are represented in each grid-cell.

#### 3.1.1.2 ISIMIP

A comprehensive review of the ISIMIP crop models, including their input data and key characteristics and differences in their modelling approaches, can be found in Rosenzweig et al. (2014). A brief description of the models along with representative description papers is given here.

Three of the seven models (EPIC, GEPIC and pDSSAT) are site-based crop models, a category of crop models developed for field-scale modelling and representation of dynamic feedbacks between the crops, soil and the atmosphere and the interactions with crop management practices (Rosenzweig et al., 2014). EPIC (Environment Policy Integrated Climate, Izaurralde et al., 2006) is a agro-ecosystem model that simulates crop growth, considering crop rotation and agricultural management options and was originally developed as a tool for the assessment of erosion effects on soil productivity. GEPIC (Geographic Information System (GIS)-based Environmental Policy Integrated Climate Model, Liu et al., 2007) is the GIS-based version of EPIC. EPIC and GEPIC employ the same core model but for the ISIMIP experiment, different soil parameters and management options were used, and these differences cause variations in the final results of the two models. Similar to the philosophy of EPIC and GEPIC, pDSSAT (parallel Decision Support System for Agro-technology Transfer, Elliott et al., 2014; Jones et al., 2003) simulates cropping systems as growth, development and yield of the crops in the system along with the interactions with crop management, soil, water and carbon.

Three other models (LPJ-GUESS, LPJmL and PEGASUS) belong to the family of agro-ecosystem models, which focuses on the simulation of carbon, nitrogen, energy and soil water fluxes (Rosenzweig et al., 2014). Both LPJ-GUESS (Lund-Potsdam-Jena General Ecosystem Simulator with Managed Land, Smith et al., 2001) and LPJmL (Lund-Potsdam-Jena managed Land Dynamic Global Vegetation and Water Balance Model, Fader et al., 2010) are dynamic vegetation models. Representation of crops was included in these models to improve the dynamic water and carbon cycle simulation. PEGASUS (Predicting Ecosystem Goods And Services Using Scenarios model, Deryng et al., 2011) simulates crop carbon dynamics coupled with surface energy and soil water exchange schemes, based on crop management data. Finally, IMAGE (Integrated Model to Assess the Global Environmental Change, Bouwman et al., 2006) is an agro-ecological model initially developed as an Integrated Assessment Model for climate change impacts. In later versions, the agro-ecological zone methodology was incorporated in the model to simulate crop development.

#### 3.1.1.3 Comparison

For comparability, the ISIMIP and IAP2 simulations use climate data from the GFDL-ESM2M global climate model focused on the RCP8.5 emissions scenario, although the GFDL-ESM2M climate data within the IAP2 are downscaled using the RCA4 regional climate model. In order to focus on the response of the crop yield models to climate-only perturbations, the socio-economic inputs were maintained at the baseline values. The comparison was performed across different spatial scales, starting from the European scale and focusing on the regional and national level. Other assumptions were made about sowing and harvesting dates. In the IAP2, sowing and harvest dates were fixed as a characteristic of each crop (Audsley et al. 2015). Within the ISIMIP models, different approaches were taken, with EPIC, LPJ-GUESS, LPJmL and pDSSAT having a fixed planting window and GEPIC, IMAGE and PEGASUS having a dynamic sowing window, which allows adaptation to climate change (Rosenzweig et al. 2014).

A distinctive feature of the IAP2 compared to the ISIMIP models, is that it includes autonomous adaptation to satisfy the supply-demand balance so that it allocates crops to the most profitable and climatically appropriate regions to meet demand, rather than growing them in every grid-cell of the domain (even climatically unsuitable cells), as with the ISIMIP models. To investigate the effect of this feature in the simulations, IAP2 yield outputs were first compared to ISIMIP outputs for the whole ISIMIP domain and secondly to ISIMIP outputs for those grid-cells corresponding to the areas where the IAP2 selects the crop (non-zero crop yield mask).

#### 3.1.2. Results

Projected changes in simulated wheat, soy and rapeseed yield in the 2050s relative to the baseline averaged over sub-regions of Europe are shown in Figure 2. Three key features can be observed in this figure. First, when all the grid cells of the domain are included in the calculations, substantial variations between the ISIMIP models and between ISIMIP models and IAP2 can be observed. However, when the non-zero IAP2 mask is imposed on the ISIMIP domain, removing those parts of Europe which the IAP2 considers to be climatically unsuitable or less profitable (due to lower productivity than competing areas of Europe), the distribution of the ISIMIP models' yields changes, with the median yields moving closer to the IAP2 values and with less zero or very low yields. It is also interesting to note that the distributions of the ISIMIP models more closely resemble each other after the ISIMIP non-zero mask is applied. These results suggest that the IAP2's selection of the most climatically uniform (and agronomically unrealistic) approach adopted by the global ISIMIP models. Lacking this selective feature, ISIMIP models are forced to simulate crops in all the grid-boxes of their domain, resulting in misleading very low yields in the areas that are not suitable for a particular crop.

Second, these effects are more pronounced for crops that are grown less across Europe (grain maize and soy), which have more constrained climatically-suitable conditions compared with more ubiquitous crops such as wheat. Finally, there appears to be a systematic difference between IAP2 and the more limited set of ISIMIP models for the simulated yields for oilseed rape, with the maximum yield within the ISIMIP domain being around 5 t/ha. Whilst the Eurostat databases do not contain any yield information for oilseed rape, the average reported yield across 17 UK sites from 2014 to 2017 is 5.48 t/ha, which is included in the IAP2's distribution, but cannot be captured by either of the two ISIMIP models for which rapeseed yield simulations are available.



Figure 2: Box and whisker plots for crop yield values across Europe, for the ISIMIP models and IAP2. (a) Unmasked: all the grid cells of the ISIMIP models' domain are included, (b) Masked: only the gridboxes where IAP2 has non-zero values are selected from the ISIMIP models' domain. Results are shown for four crops: wheat, maize, soy and rapeseed. Boxes represent the inter-quartile range and the median and whiskers the minimum and maximum crop yield values.

A comparison between ISIMIP and IAP2 modelled yields with Eurostat wheat and soy yield data at the country level (Figure 3) provides further insights into benchmarking of model performance. For most countries, the national-average Eurostat wheat and soy yields are captured by the minimum to maximum range of national-average yields of the ISIMIP models, which is, however, quite broad. The IAP2 shows a good performance in simulating wheat yields for most countries, yet consistently underestimates the national-average soy yield. As IAP2 has been calibrated against observed crop yields in Europe, its results are better for wheat - a crop that is widespread in Europe. In contrast, the IAP2's performance for soy is likely to be constrained because of its limited production area in Europe meaning that a proper calibration of the model was more difficult to achieve. Thus, in the case of specialist crops that are rarely grown in Europe, the globally calibrated ISIMIP models appear to give a better representation of the yield.



# Figure 3: Country-level average annual yields of (a) wheat [t/ha] and (b) soy [t/ha] for the baseline period, for the range of the seven ISIMIP models, IAP2 and crop yield data from Eurostat. Results are shown only for the countries for which Eurostat data were available (i.e. not including the UK).

Projected changes in simulated wheat, soy and rapeseed yield in the 2050s compared to baseline, averaged over sub-regions of Europe are shown in Figure 4. The ISIMIP data shown have been masked based on the non-zero IAP2 domain. For all the crops, there is a spatial variability in the magnitude and direction of the projected changes between models, which is not always reflected in the change calculated at the pan-European scale. While there is not always an agreement on the sign of the projected yield change between the ISIMIP models, the IAP2 generally projects the same direction of change as the majority of ISIMIP models.



Figure 4: Regionally averaged absolute changes in (a) wheat, (b) soy and (c) rapeseed yield in the 2050s under RCP8.5 compared to baseline, for the ISIMIP models and IAP2 for the eight European sub-regions defined in Figure 1.

#### 3.2. Forest productivity

For the forestry sector, simulation studies were conducted independently for the different case studies and scales. These analyses of the impact of climate change highlighted that both the continental-scale patterns and local specificities were complementary and highly useful to assess future forest management and adaptation strategies.

#### 3.2.1. Inter-model comparison

Forest productivity was modelled using two related models: ForClim and LandClim. ForClim (Bugmann and Solomon, 2000; Shao et al., 2001) was used to simulate forest development and productivity, on a 5km grid covering Scotland, for different management (i.e. different tree species) and climate change scenarios. ForClim is a cohort-based, stand scale, dynamic vegetation model that was developed to analyse successional pathways of various forest types (Bugmann and Solomon, 2000; Shao et al., 2001). Based on the theory of patch dynamics, tree development (growth), establishment and mortality are

simulated with an annual time step for small areas ("patches"), while the influence of climate and ecological processes is taken into consideration using a minimum of ecological assumptions. No interaction is assumed between trees of adjacent patches, i.e. the successional pattern at larger scales (forest stand to landscape) is obtained by averaging the simulation results from many patches (Bugmann, 2001). ForClim is composed of four independent sub-models for weather (computes relevant bioclimatic variables), water (computes an annual site-specific drought index), plant (calculates establishment, growth and mortality of trees on the forest patch) and management by simulating several cutting/harvesting and thinning techniques defined by the type (e.g. clear cutting, 'plentering'), and the frequency and intensity of management operations.

The spatially-explicit landscape forest model LandClim (v1.6) (Schumacher et al., 2004) was used in the Iberian case study at the scale of the entire Tagus river basin to simulate the response of tree-based, land uses (Montado, pine and olive plantations, and natural forests) under various climate change scenarios, fire regimes, and management strategies. LandClim is a stochastic forest landscape model designed to simulate long-term forest dynamics and the impact of climate, disturbances (i.e. fire, wind, bark beetles) and management on a wide range of ecosystem services. The model is spatially-explicit and represents the landscape on a 25 m x 25 m grid. Within each grid cell, vegetation dynamics are represented using a simplified forest gap model where different cohorts represent trees of the same species and age (Schumacher et al., 2004). Processes such as establishment, growth, mortality and competition for light and water are modelled explicitly as being driven by temperature, precipitation, soil properties and topography. Establishment and mortality are stochastic processes, thus the environment influences the probability of an event, but the event itself is determined by a uniform random number generator. Spatially-explicit processes such as seed dispersal, disturbances and management connect individual grid cells. By using different time steps for different processes, LandClim can efficiently simulate large landscapes while maintaining a relatively high degree of detail at the local scale. The model has been used to simulate a variety of forest types and forest processes in Central Europe (e.g. Schumacher et al., 2006; Temperli et al., 2012; Elkin et al., 2013) and the Mediterranean (Henne et al., 2013; Henne et al., 2015), producing results that were consistent with empirical data.

Cross-scale simulation results between ForClim and LandClim were not directly comparable because the simulations were based on distinctly different assumptions, including species choice, forest management regime, and the output variables that were considered (as described above). Yet, the simulations across different scales provide complementary information and, together, a rich picture of the future of European forests arises.

#### 3.2.2. Results

At the continental scale, climate change is projected to have positive impacts on forest productivity in cold-limited and water-unlimited areas, and rather negative impacts in areas that are or will be water-limited. Productivity maps for Scots pine (*Pinus sylvestris*) serve to illustrate this pattern at the European scale (Figure 5), with enhanced productivity projected under all climate change scenarios in Northern Europe and in the mountain areas of Central Europe, while productivity decreases in Southern and Eastern Europe, although the severity and extent of the negative impacts depends strongly on the severity of climate change.

These projections are consistent with those made in the regional case studies, although the set of species is different. In Scotland, the regional model application simulates an increase in timber production in the Scottish Highlands (Figure 6), but this is limited to the highest elevations for Scots pine, whereas it is also projected to occur at lower elevations for Sitka spruce (*Picea sitchensis*) and Douglas-fir (*Pseudotsuga menziesii*) (results not shown). The two latter (exotic) species may thus be considered as good candidate species for future afforestation in Scotland.



Figure 5: Potential productivity of Scots pine under current climate conditions (left) and projected change by 2100 under two climate scenarios (right). The intermediate climate change scenario (middle) refers to the HadGEM2-ES-RCA4 climate model for RCP4.5 and the high-end scenario to HadGEM2-ES-RCA4 for RCP8.5. Negative values indicate a loss compared to current climate, and positive values show an increase compared to current climate.



Figure 6: Harvested volume of Scots pine grown under current climate conditions (left) and under three climate scenarios (right). The total harvested volume is shown for current climate, and the absolute change in volume is shown under future climate change. Negative values (purple) indicate a decrease in volume and positive values (green) indicate increases in volume in the future. The low-end climate change scenario refers to the GFDL-ESM2M/RCA4 climate model for RCP4.5. The intermediate climate change scenario (middle) refers to HadGEM2-ES-RCA4 for RCP4.5 and the high-end scenario to HadGEM2-ES-RCA4 for RCP8.5.

In the Iberian case study, the model projects mostly negative effects that, combined with high grazing pressure (i.e. cattle in the Montado agro-forest system), leads to significant losses in cork production, especially under high-end climate change (Figure 7). These results (cork production) cannot be compared directly to the general forest productivity results from the continental-scale analysis, but they at least agree in their direction of change. In this case, maintaining ecosystem service provisioning at current levels would only be possible under low or intermediate climate change, with a reduction in grazing pressure to historic levels.



Figure 7: Projections of future decadal cork production (from Cork oak, Quercus suber) under three climate change scenarios. The first scenario assumes constant climatic conditions similar to baseline (black), the second (orange, top) represents an intermediate climate change scenario (GFDL-ESM2M-RCA4 for RCP4.5) and the third (blue, below) represents a high-end climate change scenario (HadGEM2-ES-RCA4 for RCP8.5).

#### 3.3. Water

#### 3.3.1. Inter-model comparison

Several hydrological and water resource models were applied in IMPRESSIONS, in particular:

- CFFlood, simulating coastal and fluvial flooding;
- WaterGAP meta model (WGMM), simulating changes in water resources and water use, and consequential water stress levels;
- SWIM model, a process-based catchment scale model used for the simulation of daily river discharge rates.

The results of the SWIM model can, in principle, be compared with both CFFlood and the WaterGAP meta model. However, because the SWIM model was not used for flood simulations in IMPRESSIONS, the model intercomparison was only possible between SWIM and WGMM. Furthermore, because of the spatial resolution of WGMM, the model intercomparison was only possible for three out of seven river basins, simulated with the SWIM Model.

A short introduction to the two models and results from the model intercomparison are given below.

#### 3.3.1.1 WaterGAP meta model

The WaterGAP Meta Model (WGMM) was used in the IAP2 and rIAM to assess both the impact of climate change on water resources and the change in water demand for human use due to socioeconomic development. WGMM is designed to be an emulator for the global hydrological model WaterGAP3 (Water – Global Assessment and Prognosis) (Verzano, 2009). Originally, WaterGAP3 operates on a five arc minute grid with daily internal time steps. In order to reduce runtime considerably, the spatial detail of WGMM was reduced from more than 180,000 grid cells for Europe in WaterGAP3 to spatial units with an area larger than 10,000 km<sup>2</sup>. These spatial units, hereafter referred to as river basins, are made up either by single large river basins (split into three sub-basins for the Danube) or clusters of smaller, neighbouring river basins with similar hydro-geographic properties. Moreover, the input data requirements are largely reduced as long-term statistics over a 30-year period are computed instead of daily or monthly time series.

For each river basin, the meta-model computes the change in long-term (30 years) average water availability (WA), resulting from changes in mean annual precipitation and air temperature compared to a baseline value. The meta-model relies on look-up tables populated with simulated WA of pre-run WaterGAP3 simulations for the baseline period (IAP2: CRU data set 1971-2000 [Mitchell & Jones, 2005]; rIAM: WATCH WFDEI 1981-2010 [Weedon et al., 2014]) with simultaneously modified mean temperature and precipitation.

#### 3.3.1.2 Soil and Water Integrated Model

The process-based hydrological model Soil and Water Integrated Model (SWIM) was applied to seven representative river basins in Europe: the Lule, Northern Dvina, Tay, Emån, Rhine, Danube and Tagus, as presented in Figure 8. The basins were selected in order to cover geographic and climatic heterogeneity of the European regions. The selected basins vary significantly in catchment area, climate conditions, as well as the main anthropogenic influences and associated alteration of river flow. The SWIM model was set up, calibrated and validated for all seven basins using the WATCH Era Interim dataset (Weedon et al., 2014) as climate input in the historical period. The river discharge in

the reference and future periods was simulated with the SWIM model driven by the coupled GCM-RCM climate projections, described in detail in Deliverable D2.3 (Madsen et al., 2016). The climate projections were bias-corrected to the WATCH Era Interim dataset. A fuller description of SWIM model is provided in Deliverable D3B.1 (Holman et al., 2015) for the European Case study, as well as in Deliverable D3C.1 (Rounsevell et al., 2015). The Tay, Tagus and Danube river basins (Lobanova et al., 2016, 2017) provide a link to the IMPRESSIONS regional case studies (Scottish, Iberian and Hungarian, respectively).

The SWIM model was able to adequately represent the hydrology of the river catchments to which it was applied. To judge model performance, commonly used criteria were applied – the Nash-Sutcliffe efficiency (NSE; Nash & Sutcliffe, 1970) and the Relative Volume Error (RVE). The values of RVE and NSE obtained in the calibration and validation periods for each of the basins are presented in Deliverable D3B.2 (Holman et al., 2017).



#### Figure 8: The seven river basins modelled using SWIM.

#### 3.3.2. Results

#### 3.3.2.1 Average discharge conditions

In order to test the simplified approach, implemented in WGMM, the results of the changes in water availability, simulated by WGMM, were compared to the results of the SWIM model for the basins of the Danube, the Rhine and the Tagus rivers. Figure 9 presents scatter plots for the three basins showing that the changes in the long-term average river discharge in future periods compared to the reference period from both models are in good agreement (Tagus: R<sup>2</sup>=0.66, Rhine: R<sup>2</sup>=0.76, Danube: R<sup>2</sup>= 0.77). The data in the plot covers all available climate projections (GCM-RCM identified by symbol) based on RCP4.5 (identified by blue colour) and RCP8.5 (red colour) for the periods 2011-2040 (2020s), 2041-2070 (2050s) and 2071-2100 (2080s), where the time slice is identified by symbol size. The largest uncertainty due to climate modelling was observed for the Tagus under RCP8.5, and the largest deviations of WGMM compared to SWIM results were also found for the Tagus River, but under RCP4.5 (see Deliverable D3B.2 [Holman et al., 2017] for quantitative assessment).





Figure 9: Scatter plot of relationships between changes in the average river discharge in the future periods compared to the reference period modelled by SWIM vs WGMM for the Danube, Rhine and Tagus rivers.

#### 3.3.2.2 Median annual flood discharge

A lower degree of agreement between the WGMM and SWIM was found when looking at the median annual flood discharge (QMED) (Figure 10). The correlation coefficients in the cases of the Rhine ( $R^2$ =0.29) and the Tagus ( $R^2$ =0.24) are low. The estimated changes of QMED from WGMM are systematically lower compared to those of SWIM for the Rhine. For the Tagus, WGMM mainly projects a decrease in QMED while SWIM also shows strong increases.



Figure 10: Scatter plot of change in median annual flood discharge (QMED) modelled by SWIM vs WGMM for the Danube, Rhine and Tagus rivers.

#### 3.3.3. Conclusions

Good agreement between the WGMM and SWIM Models indicates that the river basin specific 2D response surfaces within WGMM provide a suitable approach when looking at the average hydrological conditions in the river basins. The poor agreement found for the median annual flood discharge, QMED, may be explained by the fact that WGMM does not employ real climate data time series, but is based on the simple assumption that precipitation changes by a constant factor in each precipitation event. This constant factor is derived as the ratio of long-term mean annual precipitation in the scenario and the baseline period. For example, in the Tagus river basin, a decline in mean annual precipitation is assumed in WGMM, which also leads to decreasing QMED. By contrast, SWIM projects an increase in the QMED under the RCP4.5 scenario. The SWIM model takes into account the simulated precipitation pattern and dynamics, which may also bias the simulation results, especially when looking at the multi-model means. This is apparent for the RCP4.5 scenario where the multi-model spread for the first and second future time slices for the Tagus river basin is relatively large, as one model projects strong increases in the discharge and two predict strong decreases. This affects the multi-model mean and can explain the projected increase in the QMED (for mean annual monthly discharge projections of the Tagus river, see D3B.2 [Holman et al., 2017]). On the other hand, when conducting the model intercomparison of climate change impact assessment results, it is worth assessing the performance of both models on the observed dynamics of the river basins under consideration. A comparison of the WGMM and SWIM models for the historical period representation was not conducted in the IMPRESSIONS project.

Previously, the performance in the calibration and validation period for several river basins of the SWIM model and the "parent" model of the WGMM (i.e. WaterGAP3 model) were compared within the framework of the ISIMIP Inter-sectoral Model Intercomparison Project. Huang et al. (2016) conducted a comparison of the performance of regional hydrological models for 12 river basins worldwide. The intercomparison exercise also included the WaterGAP3 regionalised version of the model. In general, the WaterGAP3 model was less accurate in reproducing the average discharge conditions in the basins than the regional models. On the other hand, the reproduction of the low and high flow conditions was difficult for all models applied to the river basins, and to properly assess these, each model has to be calibrated for extreme conditions.

In general, the simulations of climate change impacts from the SWIM model might be more credible than those of WGMM, as the SWIM model is calibrated and validated for each river basin and employs real climate variables data series; on the one hand, this helps to preserve the physicality of the simulation even if it can, on the other hand, generate additional uncertainties (e.g. due to model parametrization and equifinality effects). However, when aiming at the average flow conditions, WGMM can be helpful, as it requires vastly less computational time and lower effort for set up and calibration of the model than a model of the process-based type, such as SWIM.

#### 3.4. Land use

#### 3.4.1. Inter-model comparison

Two distinct approaches to modelling land use change were undertaken in the IMPRESSIONS project. One relies on an established economic-based model that allocates land uses on the basis of their relative profitability, given the levels of demand for the goods they produce (the SFARMOD model, integrated in the IAP2) (Harrison et al., 2015; 2016). The other uses a newly developed agent-based model of land manager decision-making, developed from the CRAFTY modelling framework (Murray-

Rust et al., 2014; Brown et al., 2016a; Blanco et al., 2017). The former approach constrains production levels to meet demand levels using a pseudo-optimal land use configuration, while the latter allows the land use configuration to emerge from individual-level decision-making without necessarily meeting demand levels. As such, these models represent distinct paradigms for modelling land system dynamics, built on 'top-down' and 'bottom-up' conceptualisation of the system respectively (Brown et al., 2016b).

This fundamental difference in modelling approach makes a comparison of model outputs particularly valuable. To facilitate this and to ensure that a fair comparison is possible, the application of CRAFTY within IMPRESSIONS (CRAFTY-EU) used IAP2 outputs for input variables wherever possible, and therefore effectively provides an alternative land use model within the same climatic, socio-economic and cross-sectoral context. IAP2 is described in section 3.1.1.1 and CRAFTY is described below, before the comparison procedure and results are presented.

#### 3.4.1.2 CRAFTY-EU

CRAFTY-EU is an application of the CRAFTY framework for agent-based modelling of land use change (Murray-Rust et al., 2014; Brown et al., 2016a; Blanco et al., 2017). The CRAFTY framework allows land use outcomes to be modelled as the result of decision-making by agents, each of which can represent an individual or multiple land managers, and which produce a range of ecosystem services. Production levels are determined by the natural productivity of the land (defined through a range of 'capitals', as described below), the intensity of land management, and agents' prioritisation of certain ecosystem services. Agents are grouped into Agent Functional Types (AFTs) (Arneth et al., 2014) on the basis of their management intensity and decision-making characteristics, such as degree of focus on profitgeneration. Variation within AFTs allows for individual differences in production levels and land management decisions. Social-networks within and between AFTs allow for the diffusion of knowledge and practices that increase production levels. Finally, a population of institutional agents can be defined to represent formal (e.g. governmental) and informal (e.g. social) interests in the land system, which can make defined interventions to try to achieve defined outcomes. CRAFTY-EU is calibrated using crop yields from the IMPRESSIONS IAP2. All necessary input data are derived from this source, ensuring the transparency and internal consistency of the implementation. This model pairing also allows socio-economic and climatic scenarios to be defined on the basis of comprehensive, crosssectoral simulations of the European land system by IAP2 that have been extensively evaluated, validated and utilised.

#### 3.4.2. Results

Both the IAP2 and CRAFTY-EU were run for the period 2010-2100 under the core IMPRESSIONS RCP x SSP scenario combinations. Following this, results for the 2080s time slice (2071-2100) were extracted and compared on the basis of full spatial maps, aggregate land use proportions (NUTS2 level) and total land use proportions (European level).

Figure 11 shows the results of an experiment in which divergent results from CRAFTY-EU and the IAP2 for the 2080s under the RCP2.6 x SSP1 integrated scenario were tested for robustness to changing food prices in CRAFTY-EU. In the experiment, CRAFTY-EU food prices were steadily increased, eventually reaching a point at which the same level of food production occurred as in the IAP2. At this point the results of the two models were more similar, but still quite distinct in the degree of heterogeneity in land use patterns, with CRAFTY-EU producing relatively heterogeneous patterns compared to the IAP2. Aside from this, the general longitudinal and latitudinal gradients of land use were similar suggesting

that, despite the differences in basic assumptions about land use decision-making, the models converge to some extent under similar economic conditions; the models converge more the higher food prices are allowed to rise in CRAFTY-EU.

Figure 12 shows results aggregated across land use classes and spatial areas (to show agricultural intensity and forestry land use within NUTS2 regions). These plots reveal systematic spatial and scenario-dependent variations in the differences between the two models' results, but also some general differences that are highlighted further in Figure 13. Of the spatial differences, the clearest are in northern and southern Europe. In both regions (but especially the north), CRAFTY-EU has considerably more forestry and extensive/marginal land uses than the IAP2, while the IAP2 has considerably more intensive agriculture. In mid-latitudes, CRAFTY-EU tends to have more intensive arable agriculture and forestry, while the IAP2 has more pastoral and extensive agriculture (although this pattern is partially reversed in eastern Europe).

Within the scenarios, there is some tendency for CRAFTY to produce more forestry in low-end climate scenarios and for the IAP2 to produce more extensive agriculture in high-end climate scenarios. The effects of socio-economic scenarios are clearer, with CRAFTY tending to produce more intensive arable agriculture than the IAP2 in SSPs 4 and 5, but the IAP2 producing far more in SSP3. The IAP2 also clearly produces more pastoral agriculture in SSP1.

These differences remain strong, and informative, at the European level where a main conclusion from the model comparison is that CRAFTY-EU tends to produce more forest, while the IAP2 tends to produce more intensive agriculture (as above, in more areas). This general difference is attributable to the IAP2's assumption that food prices will rise, within limits, so that agriculture can expand to profitably meet food demand (allowing for imports), and CRAFTY-EU's contrasting assumption that prices are constrained and more balanced across goods and services (e.g. timber), thereby allowing larger shortfalls in agricultural food production. Another clear difference is in the heterogeneity of land uses, especially under extreme scenarios; the IAP2 produces less spatial diversity due to the underlying soil-climate clusters that were used to facilitate model runtimes, while CRAFTY produces more mixed patterns, reflecting underlying heterogeneity (and freedom) in agent decision-making.

These differences have substantial implications for the provision of ecosystem services, including food, under future global change. As such, the modelling assumptions that generate these differences require considerable scrutiny, particularly in terms of (i) the representativeness of the assumptions of economic rationality in land use decision-making and the extent to which economic priority would and could be given to meeting net food demands in the European agricultural industry, and (ii) the representativeness of an assumption of actor heterogeneity, and about the strength of that heterogeneity. This particular model comparison provides a strong basis for further investigation of these issues.



'base' CRAFTY result

increasing food price

•••

d price

...

## Agent type

Intensive arable Extensive arable Intensive pasture Extensive pasture Very extensive pasture Intensive agro-forestry mosaic Extensive agro-forestry mosaic Managed forest Unmanaged forest Marginal multifunctional Unmanaged land



Figure 11: Comparison of CRAFTY-EU (top, different settings) and IAP2 (bottom) for the RCP2.6 x SSP1 scenario.

#### a) CRAFTY-EU minus IAP2



RCP4.5-SSP4

RCP8.5-SSP3

RCP8.5-SSP5







b) IAP2 minus CRAFTY-EU

**31** | Page



Figure 12: The most substantially over-represented land use type in each European NUTS2 region: (a) for CRAFTY-EU (compared to the IAP2); and (b) for the IAP2 (compared to CRAFTY-EU). Opacity is higher when the difference between the models is greater.





Figure 13: The most substantially over-represented land use types across European NUTS2 regions for each scenario. Green boxes show the mean over-representation of each type by CRAFTY-EU (compared to the IAP2), and the grey boxes show the mean over-representation of each type by the IAP2 (compared to CRAFTY-EU). Order of land use types on the x-axis is Extensive, Forestry, Intensive, Pastoral, Unmanaged, and Very extensive.

#### **3.5.** Population and urban growth

#### 3.5.1. Inter-model comparison

A range of datasets and models were used in IMPRESSIONS to reflect changes in population and/or urban growth across different spatial scales. These include:

- (i) Global-scale datasets on population change related to the SSPs (Jones & O'Neill, 2016);
- (ii) European-scale datasets for population produced by Terämä et al. (2017) for IMPRESSIONS (see Deliverable D3C.1 [Rounsevell et al., 2015] section 2.3);
- (iii) European-scale model outputs for both population and urban area from the IAP2 for Europe developed for IMPRESSIONS. In the IAP2, the Regional Urban Growth model version 2 (RUG2) is used to allocate population and urban area in future scenarios;
- (iv) European-scale model outputs for both population and urban area from the Regional Integrated Assessment Model (rIAM) for Europe developed within IMPRESSIONS. The Regional Urban Growth model version 3 (RUG3) was used to allocate population and urban area in future scenarios.
- (v) Regional-scale model outputs for both population and urban area from the IAP2 for Scotland (based on RUG2), developed in IMPRESSIONS;
- (vi) Regional-scale model outputs for both population and urban area from the ALLOCATION model developed for Hungary in IMPRESSIONS (Li et al., 2016; 2017).

More detailed model descriptions are given below.

#### 3.5.1.1 NUTS2 downscaled European population

The Wittgenstein Centre (2015) holds the most recent version of the population component of the SSP database, which provides the fundamental socio-economic datasets for the IPCC process (IIASA, 2012). These datasets include population by age, sex and education; patterns of urbanisation; and information on changes in GDP. Spatially, the data are available only at a national scale. As such, there is a need to downscale these data to a finer spatial resolution to enable it to act as inputs to the modelling within IMPRESSIONS. To address this, Terämä et al. (2017) downscaled the age-specific national population projections to spatially-explicit, sub-national administrative data at the NUTS2 level using data from Eurostat.

To do this, the downscaling process made two key assumptions. First, it was assumed that the national level distribution of population by age range (from the SSPs) was constant across all national sub-units (i.e. the NUTS2 units had the same distribution as the national). Second, it was assumed that no subnational migration took place between units. Using these two assumptions, the national-scale Wittgenstein data were applied to the NUTS2 age-pyramids from Eurostat to reflect changing population for the future SSP scenarios for 10-year time steps for the period of 2010-2100. These data were used as an input to the modelling in rIAM and ALLOCATION. In the analysis below, the Terämä et al. (2017) data were used as the reference for population data against which other datasets were compared due to the fact that it is expected to be the closest representation of the SSPs.

#### 3.5.1.2 Global SSP-related population data

Jones & O'Neil (2016) produced a second spatially-explicit dataset on population for the SSPs. However, they (i) focus on changes at a global scale; (ii) use a grid-cell based approach at a 1/8° spatial resolution (rather than NUTS2); (iii) use a gravity-based model that uses potential urban and rural attractiveness to allocate population change and (iv) extrapolate from different baseline data (2.5' gridded world population dataset for the year 2000 from CIESIN [2005] rather than Eurostat data). For more details, see Jones & O'Neil (2016).

#### 3.5.1.3 IAP2 Rural Urban Growth version 2 (RUG2) model (used for both Europe and Scotland)

The original RUG model (Reginster & Rounsevell, 2006; Rickesbusch et al., 2010) simulates urban growth as a function of changes in socio-economics and societal preferences. The full model takes into consideration local geography, travel times with existing infrastructure and city type (mono- or polycentric). However, the run-time of the full model is too long to be included in the online, interactive IAP2 and therefore a simplified meta-model was embedded within the IAP2.

The RUG2 meta-model is embedded within both the European and Scottish versions of the IAP2. Like the Terämä et al. (2017) data, the RUG1 model is focussed on the European/national scale rather than the global (as with Jones & O'Neil, 2016), and the model allocates urban area and population differently. The main differences from the Terämä et al. (2017) data are: (i) the RUG2 meta-model used within the IAP2 is a look-up-table based on the 1 km x 1 km outputs of this model summarised to the appropriate resolution (e.g. at a 10' x 10' grid for Europe and a 5 km x 5 km grid for Scotland, rather than NUTS2 and (ii) RUG2 is a regression model (Reginster & Rounsevell, 2006) driven by population, GDP/capita, urban area type (large city/rural region) and country. The model takes into consideration factors such as scenario-driven preferences for urban growth, such as the desire to live within proximity of the coast or for rural locations, and preferences for constraining urban sprawl, which are IAP2 scenario inputs.

#### 3.5.1.4 The Regional Integrated Assessment Model (rIAM, used for Europe)

The Regional Integrated Assessment Model has been developed within IMPRESSIONS. The RUG version 3 model within rIAM is based on the RUG2 model within IAP2, but expanded to (i) consider time dependency between scenarios to allow it to run on 10-yearly time steps between 2010 and 2100 with each time step building on the urban extent of the previous time step, and, in addition (ii) it is customised to the SSP scenarios in a way that the original RUG model is not; it considers a greater level of detail in terms of population characterisation and considers the age structure of the population. It does this by using the spatially downscaled NUTS 2 population data from Terämä et al. (2017) as an input.

#### 3.5.1.5 The ALLOCATION model (used for Hungary)

ALLOCATION is a new grid-cell based spatial model developed specifically for the IMPRESSIONS Hungarian case study. It uses linear regression functions to project demand for urban development based on NUTS2 level inputs from Terämä et al. (2017). These NUTS2 level demands are then distributed to 1km<sup>2</sup> cells and the expected cell-level increase of each land class is then estimated. This estimation is a function of the cell development potential based on (i) accessibility and landscape configuration; (ii) policy preferences for certain land classes (e.g. commercial/industrial) in particular regions; and (iii) policy/societal preferences for compact or sprawling urban settlements. The impact of climatic change is modelled through its projected relationship with GDP. Modifying these factors by scenario allows the Hungarian case study to assess the influence of economy, climate, demographics, residential preferences and urban planning.

#### 3.5.2. Results

For each dataset, values were compared for the baseline and for each of the four SSPs at the most appropriate comparable spatial resolution. For European datasets, the NUTS2 level was used to compare all datasets (IAP2 Europe and Scotland, rIAM, Jones and O'Neil and the spatially downscaled Terämä et al. data). The official SSP quantifications from the IIASA website downscaled by Terämä et al. (2017) were used as a reference against which the other models were compared. In addition, the IAP2 and rIAM datasets were also compared at the 10' x 10' grid-cell level (for cells shared by both models). All SSP comparisons are for the same time period 2080-2100, while baseline comparisons are for *circa* 2010.

#### 3.5.2.1 Comparison for baseline

When aggregated to a NUTS2 level all datasets have a relatively high correlation with the baseline data used by Terämä et al. (2017) (Figure 14). The global and European datasets plot very close to the 1:1 line illustrating their close match with the Terämä et al. (2017) data. The regression lines diverge slightly for regions with higher population within the Terämä et al. (2017) data with the IAP2 deviating above the line and rIAM and Jones and O'Neil (2016) datasets deviating below the line. The rIAM data are the closest to the line reflecting the fact that the Terämä et al. (2017) data were used to train the modelling within rIAM, which was not the case for either the IAP2 data (pre-dating the SSPs), which was trained on CORINE land cover data, nor for the Jones and O'Neil (2016) data, which were trained on CIESIN data from 2010 (see above). Nonetheless, all lines show very highly statistically significant correlations (R<sup>2</sup> between 0.85 and 0.97) with the trends in population used by Terämä et al. (2017). Scottish and Hungarian modelling for baseline also shows a high level of correlation with the other models, plotting very close to the regression lines for the other models.



Figure 14: NUTS2-level comparison of population (million people per NUTS2 region) showing regression lines for IAP2 Europe (blue line), Jones & O'Neil data (green) and rIAM (red) and aggregate raw data for IAP2 Scotland (gold dots) and Hungary (black dots) relative to IIASA baseline data for Europe. Dashed line is 1:1 line with IIASA data.

Looking in more detail at the scatter plots for the European-scale models (Figure 15 a-c) it is clear that there are more outliers in the IAP2 data, particularly above the line, the clearest of which is the Isle de France region of France (FR10). This region has a greater population in the IAP2 modelling than in the Wittgenstein NUTS2 downscaled data. Of the three models, the rIAM data, which is trained on the Terämä et al. (2017) data, has the greatest correlation ( $R^2 = 0.97$ ). This similarity is also apparent in the mapped data (Figure 15 d-f) with fewer NUTS2 regions in the rIAM data showing values > +/- 10% from the Terämä et al. (2017) data. The maps reveal some commonalities in spatial patterns. In general, Scandinavia and Western Europe show values lower than the downscaled Terämä et al. (2017) values, while Eastern Europe and some parts of Central Europe and the UK show higher values than Terämä et al. (2017).



Figure 15 a-c: Differences between model outputs and IIASA population data at baseline at the NUTS2 scale. Some key outliers are labelled with their NUTS2 code: FR10 = Ile de France; RO01 = North-west Romania; LT00 = Lithuania; RO02 = South-east Romania; PL31 = Lublin, Poland; BE31 = Walloon Brabant, Belgium; NL32 = North Holland, the Netherlands; UKI1 = Inner London, England; AT13 = Vienna; ES51 = Catalonia, Spain; ES52 = Valencia Community, Spain and DE40 = Brandenburg, Germany. d-f: relative change from Terämä et al. (2017) NUTS2-scale data by model.
Figure 16 shows the relationships between the rIAM and IAP2 datasets at the 10' x 10' resolution at which they are both generated. The correlation between the IAP2 data and rIAM data for population is quite good ( $R^2$  =0.75). Although the deviation between the two models may be minor for a grid cell, the low values of urban area and population make the proportion of the relative differences highly visible. Moreover, there are very few grid cells where population is allocated similarly between the two models (Figure 16 c-d). A visual comparison of the spatial patterns reveals that IAP2 and rIAM have the greatest amount of agreement in terms of area with changes <+/-1% in areas with very low population (e.g. Fennoscandia, Northern Scotland and areas of Spain).



Figure 16 a-b: Scatterplots and c-d: Maps highlighting percentage difference between rIAM and IAP2 datasets for baseline at the 10' x 10' resolution for population (a-c) and urban area (b-d).

Differences in urban areas show clearer patterns outside of the areas where there is very little population, with rIAM projecting higher values around the Alps (particularly Switzerland and Austria), on the Iberian peninsula (particularly in the West), in Ireland and in central Poland. Conversely, the IAP2 projects higher values particularly in the UK, Scandinavia, Greece, Latvia and Estonia. These differences in both urban area and population datasets can be explained by the different modelling approaches. For example, rIAM downscales population data from national to NUTS 2 data whereas the IAP2 data is projected from regressions based primarily on GDP and population.

### 3.5.2.2 Comparison for scenarios for Europe

There are relatively high correlations ( $R^2$  values between 0.70 and 0.90) between the different projections of population at the NUTS2 level in the different scenarios (Figure 17). The rIAM model trained on the Terämä et al. (2017) data shows the highest correlations (ca. > 0.87 in all scenarios); the Jones and O'Neil data (2016) varies between  $R^2$  of 0.85 and 0.86, while the IAP2 data shows the lowest correlation ranging from  $R^2$  of 0.70 to 0.80. Furthermore, while the rIAM and Jones and O'Neil (2016) data project values consistently below and above the lines (respectively), the IAP2 shows different trends depending on the scenario in question, with a trend higher than the Terämä et al. (2017) data in SSP3 and SSP4, and lower in SSP1 and SSP5. Hungary and Scotland plot very close to the 1:1 line in most scenarios, except in SSP5 where the Scottish data plots considerably below the line.

A comparison of the IAP2 and rIAM outputs for urban area and population growth at the 10' grid scale shows that there are considerable differences in the spatial patterns between the two models (Figure 18). The rIAM data, based on the Terämä et al. (2017) data, shows considerably greater population across much of Western Europe and considerably lower population in Eastern Europe in a way that reflects overall patterns of change in SSP populations. This reflects the fact that the IAP2 does not target population at a national scale in the same way as rIAM. The rIAM has pre-defined levels of population for each country from the Wittgenstein (2015) SSP data. Conversely, the IAP2 projects urban area and population change based on variables such as GDP and total population change at the European scale, and has no knowledge of how these are projected to be allocated differentially between countries.

## 3.5.2.3 Comparison for scenarios for Scotland

Two models are available for Scotland – the Scottish and European versions of the IAP2 (Figure 19). The comparison below highlights that the models follow similar trajectories with high levels of correlation for both population and urban area ( $R^2 > 0.95$ ), but that despite this, there are notable differences in the spatial pattern, which results from significant deviations from the 1:1 line. This is only the case for populated regions, as for large areas of Scotland with low populations there is very little difference between the two models (both project very low population). For more populated areas, baseline and SSP3 (where population change is none or negative) show a mix of under- and over-predictions with no clear pattern. In SSPs 1, 4 and 5, however, the Scottish version of the IAP2 projects a greater population in most areas. The visible deviation of the results of the Scottish IAP2 from that of the European IAP2 is due to the values of population and urban area being small.

## 3.5.2.4 Comparison for scenarios for Hungary

The comparison between IAP2 and the ALLOCATION model (Figure 20) shows high correlations between the two models for both population and urban area data at baseline ( $R^2 > 0.91$ ) and in all scenarios ( $R^2 > 0.87$ ). The majority of regression lines also plot very close to the 1:1 line, showing that the models not only correlate, but show generally similar values. However, in SSP3 and SSP4 there are

highly correlated trends with notable differences from the 1:1 line in the population data, despite relatively strong correlations close to the 1:1 lines in terms of urban area. A closer examination of all regression lines reveals that while extreme areas of high population project close to the regression line, there are notable differences at the low-population end, with ALLOCATION often projecting higher values than the IAP2 in the baseline, SSP1 and SSP5, but lower in SSP3 and SSP4. This is seen in the considerably lower values across the majority of Hungary in SSP3/4. These differences are likely to reflect the differences in modelling approaches used between the two models with the IAP2 population allocated in consideration of changes across the whole of Europe, while the Hungarian data is trained based on SSP-specific changes projected for Hungary itself.



Figure 17: NUTS2-level comparison of population (million people per NUTS2 region) showing regression lines for IAP2 Europe (blue line), Jones & O'Neil (2016) data (green) and rIAM (red) and aggregate raw data for IAP2 Scotland (gold dots) and Hungary (black dots) relative to IIASA SSP scenarios data for Europe. Dashed line is 1:1 line with IIASA data. Descriptive change in population in brackets after SSP label is for the 2080-2100 time period.



Figure 18: Spatial patterns of the percentage difference between rIAM and IAP2 datasets at the 10' by 10' grid cell resolution for both population and urban area for the four SSP scenarios used within IMPRESSIONS.



Figure 19: Scatterplots of a: population and c: urban area (x-axis EU IAP2; y=Scottish IAP2), and maps highlighting the percentage difference in b: population and d: urban area between the Scottish IAP2 and European IAP2, at baseline and for the SSPs. The Scottish IAP2 results are aggregated to the resolution (10') of the EU IAP2. The key for the rows b) and d) is the same as for Figure 18. Reds show Scottish IAP2 > European IAP2; Blue shows the inverse.



Figure 20: Scatterplots of a: population and c: urban area (x-axis EU IAP2; y=Hungarian ALLOCATION), and maps highlighting the percentage difference in b: population and d: urban area between the ALLOCATION and European IAP2 models, at baseline and for the SSPs at the scale of the European versions of IAP2. The key for the rows b) and d) is the same as for Figure 18. Reds show ALLOCATION>IAP2; Blue shows the inverse.

## 4. Inter-model and cross-scale comparisons using impact response surfaces

Responses to future changes in climatic and socio-economic conditions can be expected to vary between sectors and regions, reflecting differential sensitivity to these highly uncertain factors. A sensitivity analysis was conducted using a suite of IMPRESSIONS impact models (for health, agriculture, biodiversity, land use, floods and forestry) across Europe with respect to changes in key climate and socio-economic variables (Fronzek et al., in review). Results were aggregated to eight European sub-regions (see Figure 1) and plotted as scenario-neutral impact response surfaces (IRS). These depict the modelled behaviour of an impact variable in response to changes in two key explanatory variables.

The sensitivity analyses of the models were conducted for at least two input variables (climate-related and/or socio-economic) such that these variables were perturbed simultaneously over ranges defined to be wide enough to encompass projections of long-term changes for the 21<sup>st</sup> century that plausibly could occur somewhere in Europe. Aggregated model results were plotted as contoured IRS with respect to the axes of the two drivers. A summary of results from the sensitivity analyses to climate change is presented here; more details as well as sensitivity analyses to socio-economic variables are presented in Fronzek et al. (in review).

IRS for changes relative to the baseline for impact variables relevant to agriculture (i.e. crop yield, net primary production – NPP, mean river discharge and intensive agricultural land use) showed general increases in these impact variables as precipitation increased, and decreases as temperature increased (Figure 21).

The patterns of yield response of the three crop species showed large regional variation. Regional changes in NPP showed smaller differences between the eight sub-regions compared to the crop yield IRS, suggesting that the local adaptation of natural ecosystems shares characteristics regardless of the prevailing regional climate. Changes in mean river discharge were mainly affected by changes in precipitation, with decreasing precipitation reducing river flows. Although the European regions differed little in the shape of their IRS, the strength of the precipitation response varied across Europe. Changes in the proportion of intensive agricultural land use decreased for increases in precipitation and temperature in all regions except Central Europe. The decreases coincided with increases in crop yields in most regions, especially the Iberian Peninsula, France and Mediterranean (Figure 21).

The stem basal area of the five simulated tree species increased under wetter conditions in all regions, whereas the temperature optimum varied between region and species (Figure 22). Distinct temperature and precipitation thresholds can be identified from the IRS, for example a sharp decline in basal area can be seen for European beech and Sessile oak in the British Isles for warming of 6°C or more. For European beech, Sessile oak and Scots pine in the Alps, a clear threshold was found for precipitation decreases of more than 20%. Large increases in basal area of more than 40 m<sup>2</sup> ha<sup>-1</sup> were simulated for several locations at their respective optimum perturbation.

The percentage of forest land use increased for wetter and decreased for drier conditions for most regions, except in Central Europe, where decreases were simulated for nearly all perturbations (Figure 22). The patterns of changes were largely opposite to those of intensive agricultural land use (see Figure 21).



Figure 21: Example of impact response surfaces of variables relevant for agriculture and their sensitivity to changes in mean annual temperature (x-axis in °C) and precipitation (y-axis in %). From top to bottom rows: change in wheat, maize and potato yields, change in net primary production (NPP), change in mean river discharge and change in intensive agricultural land use (LU). All simulations are without the CO<sub>2</sub> effect and technological development. Hatched areas denote crop yields below 1000 kgha<sup>-1</sup> and European demand not being met for intensive agricultural land use (Fronzek et al., in review).





Figure 22: Impact response surfaces of variables related to forestry and their sensitivity to changes in mean annual temperature and precipitation. From top to bottom rows: change in basal area of European beech, Sessile oak, Holm oak, Scots pine and Norway spruce (absolute change in m<sup>2</sup> ha<sup>-1</sup>) and change in forest land use (LU) (relative change in %). Hatching shows impact outcomes that are not economically viable (basal area < 2 m<sup>2</sup> ha<sup>-1</sup>). Black crosses on maps above the columns indicate the grid cell locations for forest basal area simulations (Fronzek et al., in review).

The sensitivity to changes in precipitation and temperature was estimated as the average rate of change per 10% precipitation change and per 1°C temperature change (Fronzek et al., in review). To ensure that the climate changes considered were consistent with model projections for a given region, these were calculated only for perturbations over that portion of an IRS encompassing the 5<sup>th</sup> to 95<sup>th</sup>

percentiles of 21<sup>st</sup> century CMIP5<sup>1</sup> global climate model (GCM) projections for "high-end" RCP8.5 forcing. We used a sub-set of 38 projections (van Oldenborgh et al., 2013) to calculate period-averaged changes in annual mean temperature and precipitation for the land grid cells of the eight regions. Portions of the IRS that were analysed therefore differed slightly between regions. For the same sub-set of impact indicators, the IRS were also used to estimate impacts of the 38 GCM projections. These were calculated from the IRS by interpolating to the locations of perturbations corresponding to each climate projection.

An attempt to summarise the sensitivity of the climate-driven impact indicators shown by their IRS is given in Figure 23. This plots the median sensitivity to temperature and precipitation changes by region across all indicators (coloured points) and by indicator across all regions (black symbols). For the set of impact indicators considered, the British Isles was the region with the smallest sensitivity to both temperature and precipitation (across all indicators), whereas Central Europe had the strongest median response to temperature and Eastern Europe to precipitation. The Iberian Peninsula had the largest precipitation-to-temperature sensitivity ratio, implying a relatively stronger response to precipitation than temperature compared to other regions. The British Isles showed the lowest ratio.

The median sensitivity to temperature of indicators across the regions was lowest for the two river discharge indicators and highest for the basal area of Norway spruce (Figure 23). At the low-end of sensitivity to precipitation were intensive agricultural land use, maize and potato yields and the basal area of Scots pine, whereas the largest precipitation sensitivity was found for Norway spruce. For crop yields, wheat was around three times more sensitive to precipitation than potato and maize. Similarly, the two land use indicators showed contrasting sensitivities with forest land use being more sensitive to precipitation than agricultural land use.

The variation of sensitivities across individual indicators and regions is large for most indicators (not shown). Exceptions are the two river discharge indicators and, to a lesser extent net primary production (NPP) and agricultural land use, whose regional values clustered around a small range of sensitivities both for temperature and precipitation.

To complement the results on sensitivity across the IRS, Figure 24 depicts averaged responses for each region and climate-driven impact indicator. Responses are medians across the 38 perturbations defined by the (high-end) CMIP5 RCP8.5 GCM projections for the end of the 21<sup>st</sup> century. North-eastern Europe showed increases in yields of all crops and basal area of all tree species, whereas Central and Eastern Europe showed decreases in these indicators (Figure 24). In regions of southern Europe (Iberian Peninsula, France and Mediterranean) indicators of river discharge and stem basal area (except Holm oak) were projected to decrease, whereas crop yields increased in these regions, where it was assumed that irrigation would compensate for decreases in precipitation.

<sup>&</sup>lt;sup>1</sup> Coupled Model Inter-comparison Project Phase 5 (Taylor et al., 2012)



Figure 23: Sensitivity of 13 impact indicators to changes in temperature and precipitation averaged across ranges of perturbations defined for each region by 21<sup>st</sup> century CMIP5 RCP8.5 projections (see text). Coloured points show the medians across the indicators for each sub-region; symbols show medians across the eight regions for each indicator. Points to the right (left) of the grey line show a larger (smaller) relative change in the impact indicator per 10% precipitation change than per 1°C temperature change (Fronzek et al., in review).



Figure 24: Simulated high-end climate change impacts on indicators of relevance to agriculture, forestry and flooding across eight European sub-regions. Impacts are for the period 2070-2099 relative to 1981-2010 based on 38 global climate models simulating the RCP8.5 forcing. Changes are in percent to enable comparison between indicators, and were derived from impact response surfaces plotted in original measurement units. Values are colour-coded in three adjacent cells for each indicator and region: from left to right, respectively, the 25<sup>th</sup>, 50<sup>th</sup> (median) and 75<sup>th</sup> percentile change. A cross indicates that baseline values of crop yield are below 1000 kg ha<sup>-1</sup> or of stem basal area are below 2 m<sup>2</sup> ha<sup>-1</sup>. No value (N/A) is given if the baseline value is zero (Fronzek et al., in review).

## 5. Qualitative synthesis across case studies and models

### 5.1. Description of synthesis table

To facilitate a qualitative comparison of scenario impacts across case studies and models, synthesis tables were created to highlight directions and broad magnitude of change in impact indicators in Europe and the regional/local case studies. Synthesis tables were produced for four key sectors: land use (agricultural and urban area); crop yields (wheat, barley, maize); forestry (productivity); water (discharge, availability, flooding); and health and wellbeing (heat related mortality, Lyme disease). Data were provided from the model runs undertaken for Europe as well as each regional/local case study area. Where model data were not available for regional/local case study areas, the relevant spatial window of a European model was used. For the IAP2 modelling, the sub-regions that most closely correspond to the case studies were used; for heat stress mortality (HEET), the relevant NUTS2 regions were used. The source of the data is highlighted throughout.

Mean values of impact indicators for three separate time slices relative to baseline are shown as arrows in the synthesis tables (Tables 1 to 5), representing changes over time. These time slices are:

- Baseline period for climate: 1981-2010 (only the year 2010 for socio-economic change)
- First time slice: 2010-2040 (~2020s)
- Second time slice: 2040-2070 (~2050s)
- Third time slice: 2070-2100 (~2080s)

The ranges shown in the cells of the synthesis tables reflect the approximate minimum and maximum values across the case studies for a specific indicator, aiding comparison across the different scales of the case studies. Where available and feasible, the data from different models is portrayed as different coloured arrows within the same diagram to highlight differences and similarities.

The background colour in the cells of the synthesis tables reflects the magnitude of decreases or increases in relation to the baseline. Impact magnitude was determined using percentiles with the midpoint set to zero. Hence, decreases in an impact indicator relative to baseline are in shades of red, while increases relative to baseline are in shades of blue. The stronger the shade, the stronger the impact. It should be noted that the colour does not reflect the potential benefit or dis-benefit of an impact to society or the environment. Figure 25 presents a key for interpreting background colours across all synthesis tables.

Percentile
5
25
45
Midpoint set to 0
55
75
95

Figure 25: Colour key for synthesis tables. All synthesis tables have been colour coded in the same manner for consistency. Where suitable, changes in indicators are compared to each other.

The results for four RCP x SSP integrated scenarios are displayed according to the matrix shown below (Figure 26). The output from the SWIM model is an exception, as it only has the climate (RCPs) as input, and accordingly only has two scenarios per time slice.



Figure 26: Matrices exemplifying the arrangement of scenarios in the synthesis tables, with four integrated climate (RCP) and socio-economic (SSP) scenarios (on the left) and two climate-only (RCP) scenarios (right). Note all impacts are based on the HadGEM2-ES global climate model downscaled with the RCA4 regional climate model for both RCPs.

# 5.2. Land use (agriculture/urban)

Land use changes across Europe show some interesting, and potentially linked, patterns (Table 1). Urban areas increase under all scenarios and across all case studies, but are highest under SSP5, due to urban sprawl. The increases are very moderate in Scotland, showing comparatively little change. This is related to a difference in geographical context, as Scotland is characterised by large areas with highly dispersed populations. The increases across the whole of Europe are slightly higher than in each of the case study regions, because the most densely populated areas are outside the IMPRESSIONS case study regions (areas such as southern UK, Germany, France and the Benelux countries; see Kotzeva et al. [2016]). Data from the rIAM (orange) shows smaller increases under SSP1, SSP3 and SSP4. Increases are, however, slightly higher for SSP5. This is a reflection of the model differences outlined in Section 3.5. Inputs from the socio-economic scenarios play a more important role in rIAM than in IAP2, explaining the higher increase in urban areas under SSP5, both in relation to the other scenarios, but also the difference between the two models. The SSP5 narrative involves a much higher population growth than in any other scenario as well as little urban planning regulations, thereby also leading to greater growth in urban areas in a sprawled manner. The relatively small increases in urban areas under SSP3 are because this scenarios narrative has a decreasing population.

The outcomes and directions of change for agricultural area show a more diverse pattern of change (Table 1). The most extreme changes (both increases and decreases) are found for intensive agriculture (dark red and blue background colours). Substantial increases in intensive agriculture are seen across all case studies under SSP1. This is explained by the scenario narrative, as locally or nationally grown food (see especially the Scottish case study) becomes more important (as food imports are decreased), combined with a focus on sustainable agriculture. However, under the other scenarios, intensive agriculture in this scenario meaning that less area is needed to produce food. This is particularly pronounced in Iberia with a decrease of almost 29%. The sub-region of Eastern Europe (representing the Hungarian case study) shows slightly different trends: initial small increases under SSP4 and SSP5 are followed by decreases later in the century. Changes in Scotland are much more moderate compared to the other cases and regions. The outcomes for extensive agriculture are more consistent across all cases and scenarios showing predominantly projected decreases. The decreases are greatest in Iberia and most moderate in Scotland.

Table 1: Synthesis table showing the direction and broad magnitude of impacts for land use indicators. All indicators show percentage changes relative to baseline. Background colours reflect the magnitude of changes in percentiles and refer to changes in IAP2 indicators rather than CRAFTY or rIAM. Ranges on the y-axis show the broad range of values across one indicator. Integrated scenarios are displayed in four quadrants per indicator and case study: clockwise from the top left = RCP8.5 x SSP5; RCP8.5 x SSP3; RCP4.5 x SSP4; RCP4.5 x SSP1.

		European			Sc	otland			Iberia (IF	S Region)	Easte	rn Europ	e (IRS F	legion)
changes in urban area	6 6	6 — 0 — 6 —	<u>~</u>	6 — 0 —	-	6 - 0 -		6 — 0 — 6 —	>	6 0 6	6 0 6	*	6	*
	•	0 -	4	0	-	0	-	0 -		0	0	<b>*</b>	0 -	-
changes in intensiv	-50	30 -	*	30 - 	-	-50		30 -50	~*	-50	30	*	30 -50	2
e agricultu re	-50	-50		30 - - -50 -	-	-50	-	30 - - -50 -	4	-50	30	4	30 — -50 —	X
changes in extensiv	-16	20 -	4	20 — -• -16 —	-	- 20 -  16 -	->	20 - - -16 -	4	-16	20	*	20 — —	
e agricultu re	20	- 20 16	*	20 — -•		- 20 -  16 -	<b>→</b>	20 -16	4	20 -16	20	*	20 — - <b>1</b> 6 —	4
changes in	۲ ۳ ۹	- 3 -						3		3 -9	3	•	3 -9	
pasture area	3 -9	- 3 -						3 -		3	3		3 — -9 —	-
Percentile	5 2	5 45	Midpoint set to 0	55	75	95						IAP2 CRA rIAN	<u>2</u> FTY Л	

CRAFTY (brown arrows) shows different trends, with decreases under SSP1 for intensive agriculture and much greater decreases than the IAP2 under SSP3. For extensive agriculture, CRAFTY projects increases under SSP3 and SSP4, in contrast to the decreasing trend projected by IAP2. These increases are strong in Iberia and Eastern Europe. Under SSP3, the increases in extensive agriculture are, however, not consistent across time slices. The initial increases are followed by decreases during the last time slice. This decrease is greatest in Hungary, where between 2040 and 2070, extensive agriculture increases by 20% and then drops to only 2% relative to the baseline. The reasons for the modelling differences are outlined in Section 3.4.

## 5.3. Crop yields

Crop yield outcomes are relatively consistent across cases and scenarios for Europe, Hungary and Iberia, with decreases or minor increases observed under SSP1 and SSP3 (Table 2). The largest increases are found under SSP5 and SSP4. These increases are in line with assumptions made particularly under SSP5, as investments are made into technology to increase agricultural outputs. Hungary shows a modest increase for wheat and barley in comparison to increases in Europe, Iberia and Scotland. Only Scotland shows consistent increases in yields across all scenarios, indicating that increased temperature might be beneficial to agricultural outputs in Scotland. Maize was not included in the synthesis table for Scotland, due to its limited relevance in the country. It has to be noted, however, that two of the climate models in the Scotland. Maize yields show the most consistent increases across cases. Yields remain relatively stable under SSP1 and SSP3 while they increase strongly under SSP4 and SSP5. European maize yields under SSP5 also show the highest overall increases across all crops and cases.

The varying outcomes according to scenarios can be partly explained by the scenario narratives. Under SSP1 there is a greater focus on sustainable agriculture resulting in lower artificial inputs, which leads to a decrease in yields. Under SSP5, technological developments, particularly fossil-fuel based technologies, lead to increases in crop yields. While the increases under SSP4 can be explained by improvements in green technologies leading to agricultural innovation.

Table 2: Synthesis table showing the direction and broad magnitude of impacts for crop yields (IAP2). Unit is t/ha relative to baseline. Background colours reflect the magnitude of changes in percentiles. Ranges on the y-axis show the broad range of values across one indicator. Integrated scenarios are displayed in four quadrants per indicator and case study: clockwise from the top left = RCP8.5 x SSP5; RCP8.5 x SSP3; RCP4.5 x SSP4; RCP4.5 x SSP1.

		Eur	opean		0		Scotla	and	Iberia	a (IRS Region)	0	Hun	gary
changes	15	*	-4	-	-4	/		15	-4	-4		4	15 -4
in wheat yield	-4	*	-4	↗	15 -4	*		-4	15 -4	15		5 <b></b>	-4
changes	10	,	-4	->	10 -4	7	-	10	10	10			10
in barley yield	10	*	10 - -4 -	ݕ	10 - -4 -	^		-4	10		1	4	10 -4
changes	18 -2	1	-2 =	<b>^</b>					-2	-2			18 <u>-2</u>
yield	18 — -2 —	*	-2	↗					18	- 18	11	2	18 -2
Percentile	e 5	25	45	Midpoint set to 0	55	75	95				-		

## 5.4. Forestry

Managed forest area decreases all over Europe in the IAP2 (Table 3). Conversely, European results from the CRAFTY model shows managed forest area to remain relatively stable, with only minor increases and decreases. The exception is Eastern Europe under SSP3, where managed forest area increases by 17.7%. This reflects a preference of the models (see section 3.4); CRAFTY assigns more forest area, while IAP2 projects more agricultural area to satisfy food demand from an increasing population.

Unmanaged forest area shows an increase under almost all scenarios; only under SSP1 unmanaged forest areas decrease slightly or remain relatively stable (Table 3). This is because SSP1 has more agricultural land in order to meet food demand with lower imports and more sustainable agricultural practices, which comes at the expense of unmanaged forests. Under the remaining scenarios, agricultural area decreases, potentially resulting in re-naturalisation in the form of unmanaged forest areas. Unmanaged forest areas in Iberia under SSP3, SSP4 and SSP5 fluctuate significantly: increases are followed by decreases. This is particularly evident under SSP4, where unmanaged forest area increases to almost 20%, then decreases to 9% followed by an increase to 31% in the last time slice (values reflect absolute changes relative to baseline). CRAFTY modelling results for unmanaged forest area shows much smaller changes compared to the IAP2 results. The highest increase for the last time slice slice is 6% in Eastern Europe.

For forest productivity, changes are summarised for Iberia only as other forest productivity model outputs were not compatible with the format of the synthesis tables and so, it was not possible to compare outputs. Iberia provides an interesting example of significant changes in the forestry sector (Table 4). Oak and cork production analysed in Iberia is projected to decrease, significantly impacting traditional lifestyles and traditions of agroforestry. For a detailed exploration of the issues see Deliverable D3C.2 (Clarke et al., 2017). However, as highlighted in Section 3.2, forest productivity is expected to increase in northern Europe and mountain areas, while it is expected to decrease in southern Europe. The impacts in Scotland are slightly more mixed (not shown in synthesis table), with productivity for Douglas fir and Sitka spruce projected to increase, while the native Scots pine does not fare as well under high-end climate change. These results highlight the need to identify tree species that will be well suited to future climate change to ensure viability and resilience of forestry sectors in different parts of Europe.

**55** | Page

Table 3: Synthesis table showing the direction and broad magnitude of impacts for forest indicators from the IAP2 and CRAFTY models. Arrows show percentage change relative to baseline. Background colours reflect the magnitude of changes in percentiles. Ranges on the y-axis show the broad range of values across one indicator. Integrated scenarios are displayed in four quadrants per indicator and case study: clockwise from the top left = RCP8.5 x SSP5; RCP8. 5x SSP3; RCP4.5 x SSP4; RCP4.5 x SSP1.

	Eur	ope	Scot	tland	Iberia (IR	S Region)	Eastern Europ	e (IRS Region)
		18				18		18
Managed	-30	-30	-30	-30	-30	-30	-30	-30
Areas			18		18			18
	-30	-30	-30	-30	-30	-30	-30	-30
Unmanag	35	35	35	35	35	35	35	35
ed Forest	25	25	25	25	25	35		
Area	-6	4	-6 -	-6	-6	-5	-6	*
Percentile	5 25 45	Midpoint set to 0	75 95				IAP CR4	2 →

Table 4: Synthesis table showing the direction of impacts for tree species productivity in Iberia based on the LandClim model. Units of Indicators: Pine plantations = m3/ha; Cork oak = kg/ha. No background colours are shown as the indicators are for the same case study and the direction of change is similar in magnitude. Ranges on the y-axis show the broad range of values across one indicator. Integrated scenarios are displayed in four quadrants per indicator and case study: clockwise from the top left = RCP8.5 x SSP5; RCP8.5 x SSP3; RCP4.5 x SSP4; RCP4.5 x SSP1.

Pine Planta	tions Iberia	Cork Oak productivity Iberia				
-80	-80	-1000	-1000			
-80	-80	-1000	-1000			

## 5.5. Water

On average, water availability is projected to increase across Europe (Table 5). However, Iberia is likely to suffer from a decrease in water availability under high-end climate change due to decreases in precipitation and higher evapotranspiration, while in Scotland, water availability remains relatively constant due to increases in precipitation.

The water exploitation index (WEI), which shows water demand as a proportion of availability, increases in Europe and across the regional/local case studies, showing that a higher proportion of available water is being used (Table 5). The index only remains stable for Scotland, showing very small decreases, indicating that Scotland is not going to be vulnerable to overexploitation of water. This is a reflection of the abundance of water resources available in Scotland, increases in precipitation in the climate scenarios and the relatively low population density even under future scenarios. Increases in the index are particularly high in Iberia, which is related to projected decreases in precipitation coupled with greater evapotranspiration due to higher temperatures and greater water demands for agricultural and other purposes. Without appropriate action, Iberia as well as Eastern Europe will suffer severe water stress under SSP5. A WEI of  $\geq$  0.4 generally indicates severe water stress, where regions are vulnerable to water exploitation. A WEI of 1 or greater is generally seen as the point after which societies will be unable to cope with water stress. An in-depth exploration of regional variations in WEI across Europe is given in Deliverable D3B.2 (Holman et al., 2017).

The likelihood of people being flooded rises under SSP1 and SSP5 in Europe as well as Iberia and Eastern Europe (Table 5). In Scotland, on the other hand, flood risk remains stable, with the number of people expected to be flooded only increasing marginally. This lack of vulnerability to flooding in the Scottish case study is an important element in both the scenarios as well as the pathways. The model simulations show the highest increases are projected under the SSP5 scenario in Europe, Iberia and Eastern Europe. Decreases are projected under the SSP3 and SSP4 scenarios; these are particularly strong in Eastern Europe and strongest for the SSP3 scenario. The strong decreases under SSP3 are related to a declining population in most areas of Europe. Under the SSP5 scenario, population increase combined with higher flood risk under high-end climate change leads to more people being flooded. For further details, see Deliverable D3B.2 (Holman et al., 2017).

#### 57 | Page

Table 5: Synthesis table showing the direction and broad magnitude of impacts for water indicators. Units of Indicators: water availability = mill. m<sup>3</sup>/a; changes in people flooded = x100 persons. Background colours reflect the magnitude of change in precentiles. Ranges on the y-axis show the broad range of values across one indicator. Integrated scenarios are displayed in four quadrants per indicator and case study: clockwise from the top left = RCP8.5 x SSP5; RCP8.5 x SSP3; RCP4.5 x SSP4; RCP4.5 x SSP1.

		E	urope				Scotla	and	Ib	eria (IRS Region)	Easte	Easter Europe (IRS Re	
Change in water	7000 -		- 7000	~	7000			7000 —		7000	7000	7000	$\overline{\mathbf{x}}$
availabili ty	-6000 —		6000		-6000		-	6000	-6000	-6000	-6000	-6000	
5)	0.7 —		0.7		- 0.7			0.7 —	0.7	0.7	- 0.7	0.7 —	
Change in water	-0.1	↗	0.1	1	0.1			0.1	-0.1	-0.1	-0.1	-0.1	×
exploitat ion index	0.7 —		- 0.7		0.7		(	0.7	0.7	0.7	0.7 —	0.7 -	_
	-0.1	-	0.1	~	0.1	_	•	0.1 —	-0.1	-0.1	-0.01	-0.01	<u> </u>
Changes	7	×	- 7	-	7	-		7>	- 7		7	* -	
in people flooded	7	*	- 7 -	-	7 -			7		→ <sup>7</sup> →	7	→ 7 -	-
	-5		5		5 -			5	-5	-5	-5	-5	
Percentile	5	25	45	Midpoint set to 0	55	75	95	IAP2 rIAM	$  \longrightarrow  $				

Changes in river discharge are modelled using the SWIM model (see Section 3.3) using inputs from RCPs without socio-economic inputs from SSPs. Consequently, there are only two scenarios rather than four for each time slice. Looking at the selected river catchment areas reveals a diverse picture of impacts (Table 6). While discharge is expected to increase in the rivers Rhine and Tay for both RCP4.5 and RCP8.5, the discharge of the river Tagus will decrease significantly, almost 50% by 2100 under RCP8.5, relative to the baseline. According to the projected results, the Danube will experience a 10% decrease in discharge by the end of the century under RCP4.5. The modelling results for the Danube under RCP8.5 show a slight decrease over the first two time slices followed by an increase at the end of the century. Peak inflows into the Tomkogul reservoir of the Syr Darya catchment area, studied as part of the Central Asia (EUx) case study, are likely to increase.

High flow events are likely to increase in the rivers Rhine and Tay, especially under the RCP8.5 scenario. High flow events for the case of the river Tagus are projected to decrease slightly under RCP4.5 and significantly under RCP8.5. The Danube is expected to experience fewer high flow events under RCP4.5, while the number of high flow days is expected to increase under RCP8.5. The rivers Syr Darya and Amu Darya (of the Central Asia (EUx) case study area) show a slight increase in peak flows under RCP4.5 and a slight decrease under RCP8.5. For further details on all the water indicators, see Deliverables D3B.2 (Holman et al., 2017) and D3C.2 (Clarke et al., 2017).

Table 6: Synthesis table showing direction and broad magnitude of changes in select river catchments: data from the SWIM model. For changes in discharge the Central Asia data reflects peak inflows into the Tomkogul reservoir in the Syr Darya Catchment. For changes in flooding the Central Asia data shows the peak flows of Amu Darya and Syr Darya, as the changes are similar. The Central Asia data only shows change from baseline to 2100. All arrows show percentage change relative to baseline. Background colours reflect the magnitude of changes in precentiles. Ranges on the y-axis show the broad range of values across one indicator. Scenarios are displayed per indicator: RCP4.5 on the left, RCP8.5 on the right.

River System		Rh	ine				Тау			Tagus			
Changes in	18 _		18 -		18 — —	-	18 -	7		18 —		18	
discharge	-50 —		-50 -		-50 —		-50 -		—	-50 —		-50	
Changes in	300 —	~	300 -		300 —	-	- 300			300 —		300	
flooding	-100		-100 -		-100 —		100			-100 —		-100	
River System		Dai	nube		Amu	Darya a	nd Syr Da	arya					
Changes in	18	>	18 — —		18 — _•		18						
discharge	-50 ——		-50 —		-50 —		-50 —						
Changes in	300 —		- 300 -		5 —		5						
flooding	-100 —		-100		-5 —		-5						
Percentile	5	25	45	Midpoint set to 0	55	75	95						

### 5.6. Social and wellbeing

As highlighted in the scenario narratives, each scenario assumes a different population development. For Europe, under SSP1 there is limited population growth; under SSP5 there is strong population growth; under SSP3 population is projected to continually decrease; while under SSP4 after initial increases, the population decreases toward the end of the century. In the Hungarian case study, population is expected to decrease under all scenarios except SSP5. In the Scottish case study, population is expected to increase under all scenarios, with the exception of SSP3, where population decreases. The highest population growth is expected under SSP1.

The ecosystem services supply/demand gap has been modelled under CRAFTY. This indicator shows the proportional gap in the supply and demand of different ecosystem services. The indicator has been included among the wellbeing indicators as it analyses the continued provision of key ecosystem services crucial to human wellbeing. The supply of ecosystem services across Europe is projected to meet the demand of scenarios SSP1, SSP4 and SSP5 (Table 7). However, under SSP3 a shortfall is projected. This is in line with the European scenario storyline for SSP3, which imagines a Europe that relies on significant resource inputs to keep the economy growing. This leads to environmental degradation and a decline in ecosystem service provision.

The potential spread of Lyme disease under high-end climate change was analysed for the Scottish and Hungarian case studies (Table 7). This found that the number of infectious ticks per km<sup>2</sup> will likely increase in both cases, as rising temperatures have a positive impact on tick populations as well as extending the period when ticks are actively questing for new hosts. This causes more frequent tickhost contacts to take place over an extended period of time, thereby increasing the likelihood of Lyme disease spreading to the human population. The increases are particularly strong in Scotland (see Deliverable D3C.2 [Clarke et al., 2017]). The increased risk of Lyme disease in these areas necessitates that health practitioners and people at risk are aware of what to do in the case of tick bites, as well as how to recognise the symptoms of Lyme disease.

Heat mortality was calculated across Europe for three different age groups (0-64, 65-74 and  $\geq$ 75), as well as considering the impacts of adaptation measures on mortality rates (Table 8). As highlighted in Deliverable D3B.2 (Holman et al., 2017), adaptation measures only appear to have a moderate effect on the number of heat related deaths (see Table 9 for impacts with no adaptation measures). Adaptation measures include improvements in housing design and urban planning, and heat wave plans with a particular focus on older and vulnerable citizens. Not surprisingly, the most severely impacted age group across Europe are people over the age of 75. For this age group, increases are highest for scenarios SSP5 and SSP3, which are associated with the high-end climate scenario of RCP8.5. The highest increases take place in Iberia, while in Scotland heat related mortality remains relatively stable. This is in line with the lower temperature increases projected to take place in Scotland relative to the rest of Europe. The development of heat mortality rates is dependent on population development and particularly on population ageing. A detailed exploration of this topic can be found in Deliverable D3B.2 (Holman et al., 2017).

Table 7: Synthesis table showing the direction and broad magnitude of impacts for human wellbeing indicators. Lyme disease risk, based on the LYR model, shows the change in the number of infectious questing nymphal ticks per km<sup>2</sup>. The ecosystem services supply shows the proportional gap across different services; Values are normalized so that numbers >1 show overproduction and numbers <1 reflect shortfall. Background colours reflect the magnitude of changes in precentiles. Ranges on the y-axis show the broad range of values across one indicator. Integrated scenarios are displayed in four quadrants per indicator and case study: clockwise from the top left = RCP8.5 x SSP5; RCP8.5 x SSP3; RCP4.5 x SSP1.

	Scot	land	Hun	gary		
	140000 ———	140000	140000	140000 ———		
Changes in Lyme	0		0	0		
disease	140000 ———	140000	140000 ———	140000 ———		
risk						
	0	0	0	0		Percentile
	Fur	one				5
_	Eur	ope				5 25
Change	2.4	оре 2.4 ————				5 25 45
Change in	2.4	<b>ope</b> 2.4				5 25 45 Midpoint set to 0
Change in ecosyste	2.4	ope 2.4				5 25 45 Midpoint set to 0 55
Change in ecosyste m services	2.4	ope 2.4 0				5 25 45 Midpoint set to 0 55 75
Change in ecosyste m services supply/d	2.4	0 2.4 0 2.4 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0				5 25 45 Midpoint set to 0 55 75 95
Change in ecosyste m services supply/d emand	2.4	<b>ope</b> 2.4 0 2.4 2.4		· · · · · · · · · · · · · · · · · · ·	LYR	5 25 45 Midpoint set to 0 55 75 95 95

#### **61** | Page

Table 8: Synthesis table showing the direction and broad magnitude of impacts of the integrated scenarios on heat mortality with adaptation measures. Data from the HEET model. Note that regions correspond to European NUTS2 regions. Background colours reflect the magnitude of changes in precentiles. Ranges on the y-axis show the broad range of values across one indicator. Integrated scenarios are displayed in four quadrants per indicator and case study: clockwise from the top left = RCP8.5 x SSP5; RCP8.5 x SSP3; RCP4.5 x SSP4; RCP4.5 x SSP1.

	Eur	оре	Scot	tland	Ibe	eria	Hun	gary
llest	20	20	20	20	20	20	20	20
neat mortality	-120	-120	-120	-120	-120	-120	-120	-120
under 64a	20	20	20	20	20	20	20	20
	-120	-120	-120	-120	-120	-120	-120	-120
	90	90	90	90	90	90	90	90
Heat Mortality	-80	-80	-80	-80	-80	-80	-80	-80
65-74s	90	90	90	90	90	90	90	90
	-80	-80	-80	-80	-80	-80	-80	-80
	2000	2000	2000	2000	2000	2000	2000	2000
Heat Mortality	0	0	0 -	0	0	0	0	0
over 75s	2000	2000	2000	2000	2000	2000	2000	2000
ļ	0	0	0 -	0	0	0	0	0
Percentile	5 25 45	Midpoint 55 set to 0	75 95					

Table 9: Synthesis table showing the direction and broad magnitude of impacts for heat mortality with no adaptation measures. Data from the HEET model. Note that regions correspond to European NUTS2 regions. Background colours reflect the magnitude of changes in precentiles. Ranges on the y-axis show the broad range of values across one indicator. Integrated scenarios are displayed in four quadrants per indicator and case study: clockwise from the top left = RCP8.5 x SSP5; RCP8.5 x SSP3; RCP4.5 x SSP4; RCP4.5 x SSP1.

	Eur	ope	Scot	tland	Ibo	eria	Hun	gary
Heat	30	30	30	30	30	30	30	30
mortality	-100	-100	-100	-100	-100	-100	-100	-100
under 64a	30	30	30	30	30	30	30	30
	-100	-100	-100	-100	-100	-100	-100	-100
	110	110	110	110	110		110	110
Heat Mortality	-80	-80	-80	-80	-80	-80	-80	-80
65-74s		110	110	110	110	110	110	110
	-80	-80	-80	-80	-80	-80	-80	-80
llest	2400	2400	2400	2400	2400	2400	2400	2400
neat Mortality	-200	-200	-200	-200	-200	-200	-200	-200
over 75s	2400	2400	2400 ———	2400	2400	2400	2400	2400
	-200	-200	-200	-200	-200	-200	-200	-200
Percentile	5 25	45 Midpoint set to 0	5 75 95					

# 6. The Central Asia (EU external) case study

### 6.1. Background

The EU External (EUx) case study differs from the other IMPRESSIONS case studies in Europe in two main ways:

- While the other case studies have investigated the direct impacts of climate change on the case study regions themselves, the objective of the EUx case study was to rethink EU strategies towards Central Asia in light of high-end socio-economic and climate change scenarios, taking into consideration geopolitical dynamics with Russia and China. Hence, the interest was to understand the implications of direct climate change impacts in one region Central Asia<sup>2</sup> for the EU, to whom these can be regarded as indirect or "cross-border" impacts (see Deliverable D3A.2 [Benzie et al., 2017]).
- 2. As no CCIAV models existed for Central Asia within the IMPRESSIONS consortium, EUx has relied on climate change impact assessments for Central Asia from the literature in addition to new global modelling work that was carried out in IMPRESSIONS. This was achieved by mapping results from the literature to the four scenarios developed in this case study.

The Central Asia region is not well covered in many global climate and integrated assessment models. For example, it is frequently grouped together with other countries into a "former-Soviet Union" region, despite the very different socio-economic, topographical and climatic contexts of countries within this grouping. Furthermore, it has not been studied much at the regional scale in previous climate impacts, vulnerability and adaptation research. While various studies exist with a more local focus within Central Asia, because the case study looked at the region as a whole (including transboundary natural and human systems), the literature base upon which to draw for this regional case study was thinner than that which is available for other world regions.

The four SSP x RCP scenarios specifically developed for the EUx case study were:

- Utopistan (SSP1 x RCP4.5 moderate climate change)
- Regional Rivalry (SSP3 x RCP8.5 high-end climate change)
- A Game of Elites (SSP4 x RCP4.5 moderate climate change)
- Fossil-Fuelled Development (SSP5 x RCP8.5 high-end climate change)

The following two sections first summarise the direct impact of climate change in Central Asia. These were presented in the second of a series of three stakeholder workshops. Some of the impacts that have been modelled as part of the IMPRESSIONS project including elements of the stakeholder-developed scenarios as direct inputs to new impact modelling carried out in the case study. The second section attempts to relate qualitatively the impacts in Central Asia to the EU, based on discussions held at the stakeholder workshops.

## 6.2. Direct impacts of climate change in Central Asia

A synthesis of direct impacts of climate change in Central Asia is given in Tables 10 and 11 and Figures 27 a-d. Climate change impacts with substantial differences across the scenario-combinations are given in Table 10, whereas impacts common across all scenarios are given in Table 11. Interactions

<sup>&</sup>lt;sup>2</sup> The Central Asian case study region consists of the five former Soviet republics Kazakhstan, Uzbekistan, Turkmenistan, Kyrgyzstan and Tajikistan.

between the impacts under each integrated scenario is illustrated as a network diagram with icons indicating positive, negative or stable changes in impacts in Figures 27 a-d. Details of direct impacts of climate change in Central Asia for water-related variables were presented in Deliverable D3A.1 (Carter et al., 2016, Section 4.2).

Indicator	Utopistan SSP1 x RCP4.5	Regional Rivalry SSP3 x RCP8.5	A Game of Elites SSP4 x RCP4.5	Fossil-Fuelled Development SSP5 x RCP8.5
Greenhouse gas emissions*	Peak in 2040, then decline due to increased mitigation efforts.	Increase throughout the 21 <sup>st</sup> century to a high level.	Peak in 2040, then decline due to increased mitigation efforts.	Increase throughout the 21 <sup>st</sup> century to a high level.
Warming and precipitation*	Moderate warming and increases in precipitation (strongest in the north-east), but also increases in extreme heat, rainfall and drought events.	Strong warming and increases in precipitation (strongest in the north-east), but also increases in extreme heat, rainfall and drought events.	Moderate warming and increases in precipitation (strongest in the north-east), but also increases in extreme heat, rainfall and drought events.	Strong warming and increases in precipitation (strongest in the north-east), but also increases in extreme heat, rainfall and drought events.
Household water use and irrigation**	Likely to decline.	Likely to decline.	Will increase.	Remains stable until 2050 and then increases slightly, whereas irrigation water demand declines (especially from the 2050s).
Cotton and wheat production**	Could be increased, but this is highly dependent on the availability of irrigation water.	Remain far below their potential, because water for irrigation is not made available at the times when needed.	Could slightly increase as agreements on the use of water resources allow continuing some irrigation.	Becomes less important and is finally replaced by export crops that are less dependent on irrigation, wheat production increases.

Table 10: Climate change impacts for the Central Asia case study with substantial differences acros
the scenario-combinations.

\* Specific to the (RCP-based) climate projection of that scenario

\*\* Specific to the (SSP-based) socio-economic storyline in addition to the climate projection

65 | Page



Figures 27 a-d: Relationships between key elements of the Central Asia case study and indicative trends under four integrated scenarios.

Indicator	Across all scenario-combinations
Heat-related mortality**	Increased risks of heat-related human mortality and adverse effects on labour productivity.
Snow season*	Shortens, and glaciers continue to shrink.
Precipitation*	The region gets wetter in the north and drier in the south (north-east to south-west division).
Water flows*	The seasonality of water resources shifts with generally reduced summer water flows and increased year-to-year variability.

#### Table 11: Climate change impacts for the Central Asia case study with similarities across the scenariocombinations.

\* Influenced by (RCP-based) climate projection only

\*\* Influenced by the (SSP-based) socio-economic storyline in addition to the climate projection

## 6.3. Indirect impacts for the European Union

The implications of the climate change impacts in Central Asia for the EU were qualitatively explored for energy security, trade, conflict and security, and migration, taking the political and socio-economic scenario circumstances into consideration.

#### 6.3.1. Energy

EU access to Central Asian oil and gas resources was expected to decrease under the Utopistan (SSP1 x RCP4.5) scenario and increase in A Game of Elites (SSP4 x RCP4.5) and Fossil-Fuelled Development (SSP5 x RCP8.5) (Table 12). This is largely an effect of differences in energy pathways between the scenarios, where a focus on renewable energy sources is assumed in Utopistan, whereas SSP5 represents a fossil-fuel dependent world. Consequently, interdependence between the EU and Central Asia on renewable energy increases in SSP1. These trends are also affected by direct climate change impacts, for example, the positive trends in hydropower (see Figure 27) and possible increasing risks to energy infrastructure due to increases in extreme precipitation events and high river flows in the northern parts of Central Asia (which are larger under RCP8.5 than RCP4.5).

Table 12: Qualitative interpretations of cross-border effects from impacts in Central Asia to the EU for the energy sector for the four integrated scenarios in the EUx case study. Red colour indicates a decrease, and blue colour indicates an increase, with the deeper shades of colour and thicker trend lines showing a higher magnitude of change. The end of the century time slice determines the colour. The arrows are approximations of the direction of change over time.

Indicators	Utopistan (SSP1 x RCP4.5)	Regional Rivalry (SSP3 x RCP8.5)	A Game of Elites (SSP4 x RCP4.5)	Fossil-Fuelled Development (SSP5 x RCP8.5)
EU access to Central Asian oil and gas		/	_	/
EU-Central Asia renewable energy interdependence	_			

#### 6.3.2. Trade

Trade opportunities represent a significant share of the EU's interest in Central Asia, despite current low volumes and value of direct trade between the two regions. The value of EU-Central Asia trade is expected to increase under Utopistan, A Game of Elites and especially the Fossil-Fuelled Development scenario (Table 13). Agricultural exports from Central Asia to Europe are likely to rise in Utopistan, including diversified products such as fibres, fruit, vegetables and bioenergy crops. The increased efficiency of Central Asian agriculture in the Fossil-Fuelled Development scenario is also likely to increase agricultural exports to the EU. A similar pattern is expected for non-agricultural exports, with export categories collapsing under Regional Rivalry and low for the A Game of Elites scenarios.

Trade across the broader Eurasian region – for example between China, Central Asia, Russia, Southern Asia and the EU – was considered to be closely related to the building of pan-Eurasian infrastructure, as envisaged by China's Belt and Road Initiative. Both Utopistan and Fossil-Fuelled Development are conducive to major infrastructure connectivity, and therefore are expected to see big increased in pan-Eurasian trade, which opens up a number of land trade opportunities for the EU. The partial completion of Belt and Road Initiatives in A Game of Elites and their collapse under Regional Rivalry mean prospects are much lower in these scenarios. The risk of an increase in climate-related hazards on infrastructure, e.g. through extreme precipitation and floods, is larger under RCP8.5 than RCP4.5.

Table 13: Qualitative interpretations of cross-border effects from impacts in Central Asia to the EU for trade for the four integrated scenarios in the EUx case study. Red colour indicates a decrease, and blue colour indicates an increase, with the deeper shades of colour and thicker trend lines showing a higher magnitude of change. The end of the century time slice determines the colour. The arrows are approximations of the direction of change over time.

Indicators	Utopistan (SSP1 x RCP4.5)	Regional Rivalry (SSP3 x RCP8.5)	A Game of Elites (SSP4 x RCP4.5)	Fossil-Fuelled Development (SSP5 x RCP8.5)
Value of EU-Central		1		
Asia trade overall				
Agricultural exports		_		
from Central Asia to				
EU		>		
Non-agricultural				
exports from Central				
Asia to EU		▶		
Pan-Eurasian trade				/

#### 6.3.3. Conflict and security

The EU will be indirectly affected by any worsening of the security situation in Central Asia, for example by disruption to energy or commercial trade, by potential changes to migration into Europe, or by geopolitical instability that may involve or otherwise threaten the EU's interests globally. EU involvement with peacekeeping in Central Asia is expected to increase under the Regional Rivalry

scenario, in particular (Table 14). Direct engagement by the EU in Central Asian security is unlikely, but possibly foreseen under the Regional Rivalry and A Game of Elites scenarios, where EU forces may be used to prop up elite regimes. Security threats from Central Asia to Europe might increase under these two scenarios, for example if instability under Regional Rivalry leads to the radicalisation and militarisation of marginalised groups within Central Asian republics.

Table 14: Qualitative interpretations of cross-border effects from impacts in Central Asia to the EU for conflict and security for the four SSP-based scenarios in the EUx case study. Red colour indicates a decrease, and blue colour indicates an increase, with the deeper shades of colour and thicker trend lines showing a higher magnitude of change. The end of the century time slice determines the colour. The arrows are approximations of the direction of change over time.

Indicators	Utopistan (SSP1 x RCP4.5)	Regional Rivalry (SSP3 x RCP8.5)	A Game of Elites (SSP4 x RCP4.5)	Fossil-Fuelled Development (SSP5 x RCP8.5)
EU involvement in				
peacekeeping in	>			
Central Asia				-
Direct engagement				
by EU in Central		>	>	
Asia security				
Security threats				
from Central Asia to				
EU				

## 6.3.4. Migration

Current migration flows between Central Asia and the EU are small, but might increase in the future, depending on developments in both regions. Skilled migration and facilitated exchange may increase significantly under the A Game of Elites scenario, and to a lesser extent in others: population growth and improved education under Utopistan may boost this, though overall out-migration from Central Asia is projected to be low.

Forced migration was expected to increase – via illicit networks in Regional Rivalry, especially following large scale conflict mid-century – and also under A Game of Elites – but is likely to decrease or be negligible under Utopistan and Fossil-Fuelled Development.

69 | Page

Table 15: Qualitative interpretations of cross-border effects from impacts in Central Asia to the EU for migration for the four SSP-based scenarios in the EUx case study. Red colour indicates a decrease, and blue colour indicates an increase, with the deeper shades of colour and thicker trend lines showing a higher magnitude of change. The end of the century time slice determines the colour. The arrows are approximations of the direction of change over time.

Indicators	Utopistan (SSP1 x RCP4.5)	Regional Rivalry (SSP3 x RCP8.5)	A Game of Elites (SSP4 x RCP4.5)	Fossil-Fuelled Development (SSP5 x RCP8.5)
Skilled migration/				
exchanges between		>		
Central Asia and EU				
Forced migration &				
refugee movement		$\int$		
from C Asia to EU				

# 7. Discussion

This deliverable report has compared modelling results in two ways: (i) across the different scales of the case studies; and (ii) across the different types of impact models. The benefits of undertaking cross-sectoral, cross-scale and inter-model comparisons are discussed further in Sections 3-6 of this report. The cross-scale comparison identified commonalities as well as diverging trends in the projected development of certain indicators under different integrated climate and socio-economic change scenarios. The analysis showed that the magnitude, and in some cases the direction of change, differs across scale, highlighting that context-dependent solutions to climate change impacts are important. Inter-model comparisons help to provide a fuller picture of the impacts that might be expected under high-end climate change, as relying on one model alone for understanding such a complex issue as climate change impacts is limited by model uncertainty. By employing a variety of models and comparing the results, it is possible to identify areas of significant agreement between models, as well as significant model differences and hence uncertainty.

By using cross-scale as well as inter-model comparisons it was possible to create a much fuller picture of climate change impacts across Europe. Providing an overview of some of the potential future impacts under high-end climate change can enable decision-makers to undertake adaptation and mitigation efforts, despite the complexity of climate change. Modelling climate change impacts is full of uncertainties, but this study helps to provide a basis for decision-makers to overcome the barrier of the "unknowable" (Dunford et al., 2014).

## 7.1. Cross-scale synthesis of impacts

The results presented here show that the realisation of certain scenario combinations will have detrimental effects, but that their intensity will be spatially determined. Land use and crop yield indicators show strong agreement across all case studies in both direction as well as magnitude of impact. However, the magnitude of change for the Scottish case study is generally lower across all indicators than in the other regions, with the only exception being crop yields, where barley and wheat are projected to increase in Scotland across all scenarios rather than only under SSP4 and SSP5.

Forestry indicators also show general agreement across scales. In Iberia, changes over time in the unmanaged forest indicator are more varied than in the other cases. However, change at the end of the 21<sup>st</sup> Century still follows the direction of the other cases.

The synthesis of the water indicators indicates that Scotland will not experience much change in this sector. Iberia, by contrast, will be negatively impacted, particularly with regards to water availability, and the vulnerability to water exploitation will increase in this region. This is the sector with the least agreement in impacts across scales.

The heat mortality indicators also show similar developments across case studies. Within all cases, the population over 75 years of age will be most severely impacted by heat stress. This effect is strongest in Hungary. Scotland will not little change in heat related mortality.

## 7.2. Model uncertainty

It is important to consider the uncertainties associated with model analyses, in order to express the level of confidence in specific findings. This is especially important for cross-sectoral, integrated studies, which are tackling complex and sometimes poorly understood issues. There are a number of sources of uncertainty in such studies, including:

- Impact model uncertainty. This reflects the capacity of models to represent real-world processes, at an acceptable level of realism (see Section 3 for further discussion of impact model uncertainties). It also refers to how well models are able to represent adaptation processes, since such processes are underpinned by the uncertainties of human decisionmaking.
- Emissions scenario and climate model uncertainties. These uncertainties lead to uncertainties in impact model outputs that use climate data as inputs (see Sections 3-6). The uncertainties arise from the capacity of models to appropriately represent the climate system, and the uncertainty in inferring climate sensitivity.
- **Socio-economic scenario uncertainties**. These uncertainties reflect the difficulties in quantifying socio-economic parameters within scenarios (e.g. population, GDP), as well as the difficulties in describing the process interactions within narrative storylines, realistically.
- Error propagation uncertainties. In cross-sectoral, integrated models (e.g. the IAP2), errors and hence uncertainties can propagate through models from one sector to the next. This could lead to the amplification of uncertainties.
- Agent-based model uncertainties (ABMs). ABMs in particular seek to represent human behaviour and decision-making (e.g. CRAFTY), which is difficult to do given the lack of general theories of human behaviour.

There are also some specific sources of uncertainty relating to the choices made in the design of the project:

- No modelling of extreme weather events (with the exception of high flow events), which are likely to increase. This means that certain positive impacts, such as increases in crop yields or forest productivity, might be negated by extreme weather events such as droughts or extreme wet phases. Even models with a daily time-step are unable to simulate extremes if the climate scenarios used as input to the models do not account for variability.
- Uncertainties related to impacts across scales. This is of particular concern for climate change as interconnections across regions make it difficult to pin-point sources of impacts. On the

other hand, actions in one region might have unintended impacts elsewhere. To deal with this issue, the EUx case study was included to analyse to what extent the impacts of climate change in a region outside the EU and their consequences for the EU can be identified.

Recognising and highlighting uncertainties in models is important, as a lack of knowledge about uncertainties could act as a constraint to effective adaptation decisions. This does not mean that decisions are not made when outcomes are uncertain. Decision-makers are constantly faced with uncertainty, and generally are able to make decisions with uncertainty (see Deliverable D2.1 [Dzebo et al., 2015]). However, as Dunford et al. (2014) point out, it is important to inform decision-makers of the nature of these uncertainties. This enables informed decision-making on their part, and allows them to base adaptation decisions on models they choose to trust. Informing decision-makers of model uncertainties can therefore promote a proactive approach to climate change adaptation.

The quantitative inter-model and cross-scale comparison (Section 3) has highlighted a few modelspecific uncertainties that can be informed by inter-model comparisons. By using different models contemporaneously, or by using information from one model to improve another model's outputs, it was possible to identify results that have a higher degree of confidence. This is particularly important for decision-makers. This was evidenced in the case of crop yields, where the imposition of a non-zero IAP2 mask on the ISIMIP data resulted in the ISIMIP results resembling more closely the IAP2 results. This highlights how the characteristics of the ISIMIP models of allocating crops to every grid-cell instead of growing them only in the most profitable and climatically appropriate regions, as is done by the IAP2 model, can result in artificially lower crop yields.

Rather than revealing a particular issue that could be resolved by including different parameters, the comparison of water models (SWIM and WGMM) indicated that the models had strengths and weaknesses with regards to certain issues. For instance, SWIM modelling of climate change impacts might be more reliable, while WGMM produces better outputs for high river flows. This stresses the advantages of using a variety of models to account for strengths and weaknesses and the related uncertainties of different models.

In other cases, such as for the forest productivity modelling, the inter-model comparison did not serve to directly compare results across models, as the models were based on distinctly different assumptions. However, the comparative exercise did in this case serve to create a richer picture of forests in Europe under different climate scenarios, exemplifying another path for overcoming model uncertainties.

The CRAFTY-IAP2 comparison of land-use indicators not only highlighted how modelling assumptions can contribute to vastly different outcomes (thereby becoming a significant source of model uncertainty), but also highlighted the need to further investigate assumptions of economic rationale in decision-making of land use change as well as actor heterogeneity.

Differences between the rIAM and IAP2 models on population growth and the development of urban areas underline uncertainties caused by different input data. The different datasets on population used by the two models, as well as different ways of handling these data, contributed to different modelling outcomes for comparable indicators.

The use of integrated assessment models (IAMs), despite the uncertainties associated with them, provides a much fuller picture of cross-sectoral climate change impacts and highlights feedbacks between sectors. By comparison, single-sector model runs are less effective in providing outputs for interconnected sectors and less adapt at capturing the impact of socio-economic drivers that affect

multiple sectors simultaneously. They are also less able to capture feedbacks between different sectors (Harrison et al., 2016). To enable effective decision-making, the use of single-sector models (which provide detailed outputs for one specific sector, but are less able to capture cross-sectoral interconnections) in combination with IAMs is therefore preferred. In this way, sector-specific processes can be identified via single-sector models, while the IAMs provide a better systemic understanding of cross-sectoral interactions and feedbacks. In addition, and of great importance, our results demonstrate that consideration of plausible alternative socio-economic scenarios can lead to much greater differences between model outcomes than focusing on climate scenarios alone.

Furthermore, it is important to recognise the extensive testing that has gone into the modelling tools used in IMPRESSIONS, including sensitivity analysis through the IRS method. Remaining uncertainties should therefore not become a barrier to proactive decision-making on climate change mitigation and adaptation (Brown et al., 2015). The models should be seen as tools, and, as their output shows, socio-economic decisions are imperative in shaping the effects of climate change. Uncertainties aside, this result should be seen as the key driver to take climate change adaptation forward. Some of the limits of adaptation measures are highlighted in the next section.

## 7.3. Limits to adaptation

Modelling climate change impacts is important as the results (can) form the basis for decisions on adaptation measures as well as improving adaptation outcomes. However, the review by Holman et al. (2018), conducted within IMPRESSIONS, has highlighted current inadequacies in the way that adaptation is generally represented within land and water sector models. By using an integrated assessment approach, some of the barriers to adaptation often highlighted in the literature can be overcome. Adaptation to climate change is a reaction to changed circumstances, and refers to the necessary adjustments to deal with harmful, as well as beneficial, impacts of said changes (Adger et al., 2008), or "means of reducing future vulnerability to climate conditions and extremes (through reducing exposure or sensitivity or increasing coping capacity)" (Tinch et al., 2015, p. 337). It is possible to differentiate between types of adaptation: incremental change, where changes are slowly implemented; or transformative changes, which implement radical and swift change (Adger et al., 2015). These forms of change are highlighted in the discussion of pathways in Deliverable D4.2 (Hölscher et al., 2017), where adaptation pathways are differentiated from transformation pathways. When it comes to societies' adaptive capacity (the ability to carry out adaptation measures), the approach taken in IMPRESSIONS conceptualises this as a related, but distinct, concept from coping capacity, which rather focuses on the ability to deal with changes when they actually occur. In particular, adaptive capacity within the IAP2 is a function of the availability of relevant capitals (social, human, manufactured and financial) and the socio-economic scenario context, allowing adaptive capacity (and the effectiveness of adaptation) to differ between socio-economic scenarios and between different types of adaptation (e.g. technological, environmental, social). A clear distinction has practical implications by helping to "strike a balance between a focus on technological solutions and the consideration of social and economic context" as well as ensuring that sufficient attention is paid to the instantaneous coping element (Tinch et al., 2015, p. 337).

Not all adaptation measures result in immediate positive outcomes, and different adaptation measures might influence each other. Different combinations of adaptation measures, therefore, do not automatically result in an immediate amelioration of climate change impacts or a reduction in exposure. Under certain circumstances, the number of people vulnerable to climate change impacts might even increase (Jaeger et al., 2014). Vulnerability is defined as "the threat of possible future harm to human well-being" and "a society is considered to be vulnerable when the impacts are beyond its
capacity to cope" (Jaeger et al., 2014, p. 399), where this vulnerability can be conceptualised as low coping capacity<sup>3</sup>.

As highlighted in the previous section, model uncertainties should not be seen as a barrier to adaptation, but rather as opportunities for informed decision-making. This is also supported by Adger et al. (2008), who argue that uncertainty about the future and limited foresight should not be seen as a limit to what adaptation measures can be taken now. The modelling undertaken in the IMPRESSIONS project overcomes this issue by providing outcomes for different combined climate and socio-economic scenarios. This enables decision-makers to choose adaptation measures that appear robust in different scenarios. The development of pathways for the different scenarios supports such decision-making, as there are certain topical pathways that appear across scenarios. In addition, the pathways also support context-specific adaptation measures to take place. This has been explored in greater detail in Deliverable D4.2 (Hölscher et al., 2017), where the scenario-specific effectiveness of adaptation measures within the pathways to achieve a pre-defined vision was assessed using the IMPRESSIONS models. Context-specific adaptation measures are a factor that enables the overcoming of a further limit to adaptation often highlighted in the literature: an overly technological approach to adaptation that discounts social as well as cultural context (Eriksen et al., 2011).

The need to incorporate social context has been achieved in the IMPRESSIONS project by codeveloping scenario narratives with key stakeholders. The case study specific scenarios highlight different priorities when simulated by the IMPRESSIONS IAP2 dependent on the different contexts. This manifests as scenario-specific differences in the plausible implementation of adaptation options that influence the ability to meet the European vision by 2100 (Hölscher et al., 2017). This serves to highlight that adaptation decisions will be dependent on what is perceived as either a manageable risk that it is possible to adapt to or what is perceived as a change that might actually be welcome. At the same time, if a certain impact is seen as changing society to such an advanced degree that it might become unrecognisable, it is likely that a limit to adaptation will have been reached and no further action will be taken (Adger et al., 2008). In this case, rather than exhausting adaptation measures when faced by severe impacts, the perceived changes are so severe that the values and norms of the affected society are unable to deal with them, resulting in no adaptation measures being taken. The role of values and norms in determining which changes are seen as losses by society, and which are seen as opportunities, should therefore not be discounted.

## 8. Conclusions

This deliverable has presented results from a range of sectoral inter-model and cross-scale comparisons. Quantitative comparisons have been presented and analysed for a limited number of models with consistent impact indicators, and a more comprehensive qualitative comparison across models, scales and sectors has been presented in the form of synthesis tables.

<sup>&</sup>lt;sup>3</sup> The IPCC 5<sup>th</sup> Assessment Report (AR5) definition of vulnerability focuses on contextual (usually social) vulnerability. Vulnerability in IMPRESSIONS is used more along the lines of the IPCC AR4 definition, which relates to vulnerability as an outcome. The IAP2 and rIAM models include the concept of capitals that are combined into a measure of (outcome) vulnerability. In the IPCC AR5 definitions, this is defined as "risk" in which (contextual) vulnerability interacts with the hazard (climate and related environmental changes/events) and with exposure to these hazards to produce "risk".

The IMPRESSIONS case studies have advanced understanding of CCIAV for high-end scenarios and also made strides in addressing model uncertainty through the work on improving European-scale models, such as the CLIMSAVE IAP (resulting in the European and Scottish IAP2) and developing new European-scale models, such as rIAM and CRAFTY. This was supplemented by more detailed CCIAV modelling at regional and local scales using a variety of physical, statistical and agent-based models. The results from this work demonstrate the significant impacts and vulnerabilities that are likely to arise from high-end scenarios, but also some of the opportunities afforded through adaptation, land-based mitigation (bioenergy and reforestation) and transformation.

The cross-scale comparison also highlighted the benefits of looking at different scales within one project, as this can reveal important variations. Impacts are spatially and temporally variable, as demonstrated by the synthesis tables in this deliverable. Certain regions of Europe will be impacted more negatively by climate change while others might benefit. In addition, using cross-sectoral modelling approaches has also been shown to have significant benefits over single-sector models. Only by integrating different sectors can model outcomes portray the connections and feedbacks between different sectors. Integrated models are also better able to reflect socio-economic scenarios, as highlighted in the discussion because they account for indirect effects and relationships between drivers. This is particularly recognisable in the IAP2 results for agriculture and forestry.

There remains, however, much to be explored. It is important to continue testing and improving CCIAV and integrated assessment models. In addition, further case studies of regions in Europe not covered in the IMPRESSIONS project might be beneficial and reveal new insights into regional variations and priorities. In addition, the Central Asia (EUx) case study has highlighted the need to explore further cross-border impacts, both impacts the EU has on other regions in the world, but also external factors influencing climate change impacts within the EU.

## Acknowledgements

The authors would like to thank colleagues of the IMPRESSIONS project who contributed to the analysis presented in this report through discussions and presentations at five modelling workshops held in London (April 2014), Pisa (September/October 2014), Cranfield (March 2015), Copenhagen (December 2015) and Pisa (November 2016) as well as at General Assembly sessions in Oxford (January 2014), Barcelona (January 2015), Florence (January 2016), Budapest (January 2017) and Naples (January 2018).

## References

- Adger, W., Dessai, S., Goulden, M., Hulme, M., Lorenzoni, I., Nelson, D., Naess, L., Wolf, J. & Wreford, A. (2008) Are there social limits to adaptation to climate change? Climatic Change, 93, 335-354.
- Arneth, A., Brown, C. & Rounsevell, M. (2014) Erratum: Global models of human decision-making for land-based mitigation and adaptation assessment. Nature Climate Change, 4, 550-557.
- Audsley, E., Trnka, M., Sabaté, S. & Sanchez, A. (2015) Interactive modelling of land profitability to estimate European agricultural and forest land use under future scenarios of climate, socioeconomics and adaptation. Climatic Change, 128, 215-227. [DOI 10.1007/s10584-014-1164-6]

- Benzie, M., Carter, T., Groundstroem, F., Carlsen, H., Savvidou, G., Pirttioja, N., Taylor, R. & Dzebo, A. (2017) Implications for the EU of cross-border climate change impacts, EU FP7 IMPRESSIONS Project Deliverable D3A.2. Available from <u>www.impressions-project.eu/documents/1</u>
- Blanco, V., Holzhauer, S., Brown, C., et al. (2017) The effect of forest owner decision-making, climatic change and societal demands on land-use change and ecosystem service provision in Sweden. Ecosystem Services 23, 174–208. [doi: 10.1016/j.ecoser.2016.12.003]
- Bouwman, F., Kram, T. & Klein Goldewijk, K. (2006) Integrated modelling of global environmental change. An overview of IMAGE 2.4, Netherlands Environmental Assessment Agency (MNP), Bilthoven, The Netherlands. http://www.rivm.nl/bibliotheek/rapporten/500110002.pdf.
- Brown, C., Brown, E., Murray-Rust, D., Cojocaru, G., Savin, C. & Rounsevell, M. (2015) Analysing uncertainties in climate change impact assessment across sectors and scenarios. Climatic Change, 128, 293-306.
- Brown, C., Brown, K. & Rounsevell, M. (2016a) A philosophical case for process-based modelling of land use change. Modelling Earth System and Environment, 2, 50. [doi: 10.1007/s40808-016-0102-1]
- Brown, C., Holzhauer, S., Metzger, M., et al. (2016b) Land managers' behaviours modulate pathways to visions of future land systems. Regional Environmental Change. [doi: 10.1007/s10113-016-0999-y]
- Bugmann, H., Solomon, A.M. (2000) Explaining forest composition and biomass across multiple biogeographical regions. Ecological Applications, 10, 95–114.
- Bugmann, H. (2001) A review of forest gap models. Climatic Change, 51, 259-305.
- Carter, T., Rounsevell, M., Fronzek, S., Harrison, P., Holman, I. & Pirttioja, N. (2015) Integrated assessment approach. EU FP7 IMPRESSIONS Project Deliverable D3.1. Available from <a href="http://www.impressions-project.eu/documents/1">www.impressions-project.eu/documents/1</a>
- Carter, T., Fronzek, S., Alkemade, R., Holman, I., Honda, Y., Ito, A., Jäger, J., Leemans, R., Nunez, S., Oka, K., O'Neil, N., Onigkeit, J., Pedde, S., Rounsevell, M., Takahashi, K., Wimmer, F. & Yoshikawa, M. (2016) Global scale application of climate change impact, adaptation and vulnerability (CCIAV) models. EU FP7 IMPRESSIONS Project Deliverable D3A.1. Available from <u>www.impressions-project.eu/documents/1</u>
- CIESIN (Center for International Earth Science Information Network) (2005) Gridded Population of the World, Version 3 (GPWv3): Population Count Grid. Palisades, NY: NASA Socioeconomic Data and Applications Center.
- Clarke, L., Rounsevell, M., Dunn, M., Capela Lourenço, T., Tàbara, D., Pinter, L., Juhasz-Horvath, L., Gaál, Z., Holman, I., Frantzeskaki, N., Jäger, J., Christensen, J.H., Madsen, M.S., Pedde, S., Cojocaru, G., Lobanova, A., Li, S., Bugmann, H., Snell, R., Lafond, V., Harrison, P.A., Fronzek, S., Takahashi, K., Honda, Y., Ito, A., Tanaka, A., Yoshikawa, M. & Janes, V. (2017) Climate Change Impacts, Adaptation and Vulnerability Model Applications in Three Regional to Local Scale Case Studies in Europe. EU FP7 IMPRESSIONS Project Deliverable D3C.2. Available from <a href="https://www.impressions-project.eu/documents/1">www.impressions-project.eu/documents/1</a>

- Deryng, D., Sacks, W., Barford, C. & Ramankutty, N. (2011) Simulating the effects of climate and agricultural management practices on global crop yield, Global Biogeochemical Cycles, 25(2). [doi:10.1029/2009GB003765]
- Dunford, R., Harrison, P. & Rounsevell, M. (2014) Exploring scenario and model uncertainty in crosssectoral integrated assessment approaches to climate change impacts. Climatic Change, 132, .417-432.
- Dunford, R., Harrison, P.A., Jager, J., Rounsevell, M.D.A. and Tinch, R. (2015) Exploring climate change vulnerability across sectors and scenarios using indicators of impacts and coping capacity. Climatic Change, 128(3): 339-354.
- Dzebo, A., Cruz, J.M., Laurenço, T.C., Carlsen, H., Dunn, M., Cots, F., Tàbara, J.D., Juhász-Horváth L. & Pintér, L. (2015) Decision-makers needs assessment: Assessment of decision-makers' needs and capacities, drivers and barriers for using scenarios, modelling and pathways analysis. EU FP7 IMPRESSIONS Project Deliverable D1.2. Available from www.impressions-project.eu/documents/1
- Elkin, C., Gutiérrez, A.G., Leuzinger, S. et al. (2013) A 2 °C warmer world is not safe for ecosystem services in the European Alps. Global Change Biology, 19, 1827–1840 [doi: 10.1111/gcb.12156].
- Elliott, J., Kelly, D., Chryssanthacopoulos, J., Glotter, M., Jhunjhnuwala, K., Best, N., Wilde, M. & Foster, I. (2014) The parallel system for integrating impact models and sectors (pSIMS), Environmental Modelling & Software, 62, 509–516. [doi:10.1016/j.envsoft.2014.04.008]
- Eriksen, S., Aldunce, P., Bahinipati, C., Martins, R., Molefe, J., Nhemachena, C., O'Brien, K., Olorunfemi,
  F., Park, J., Sygna, L. & Ulsrud, K. (2011) When not every response to climate change is a good one:
  Identifying principles for sustainable adaptation. Climate and Development, 3, 7-20.
- Fader, M., Rost, S., Müller, C., Bondeau, A. & Gerten, D. (2010) Virtual water content of temperate cereals and maize: Present and potential future patterns. Journal of Hydrology, 384, 218–231. [doi:10.1016/j.jhydrol.2009.12.011]
- Fronzek, S., Pirttioja, N., Carter, T., Bindi, M., Hoffmann, H., Palosuo, T., Ruiz-Ramos, M., Tao, F., Trnka, M., Acutis, M., Asseng, S., Baranowski, P., Basso, B., Bodin, P., Buis, S., Cammarano, D., Deligios, P., Destain, M.-F., Dumont, B., Ewert, F., Ferrise, R., François, L., Gaiser, T., Hlavinka, P., Jacquemin, I., Kersebaum, K., Kollas, C., Krzyszczak, J., Lorite, I.J., Minet, J., Minguez, M., Montesino, M., Moriondo, M., Müller, C., Nendel, C., Öztürk, I., Perego, A., Rodríguez, A., Ruane, A., Ruget, F., Sanna, M., Semenov, M., Slawinski, C., Stratonovitch, P., Supit, I., Waha, K., Wang, E., Wu, L., Zhao, Z. & Rötter, R. (2018) Classifying multi-model wheat yield impact response surfaces showing sensitivity to temperature and precipitation change. Agricultural Systems, 159, 209–224. [doi: 10.1016/j.agsy.2017.08.004]
- Fronzek, S., Carter, T., Pirttioja, N., Alkemade, R., Audsley, E., Bugmann, H., Flörke, M., Holman, I., Honda, Y., Ito, A., Janes Bassett, V., Lafond, V., Leemans, R., Mokrech, M., Nunez, S., Sandars, D., Snell, R., Takahashi, K., Tanaka, A., Wimmer, F. & Yoshikawa, M. (In review). Comparing sensitivity to climate and socio-economic change across sectors and European regions. Submitted to Regional Environmental Change.

- Harrison, P.A., Holman, I. & Berry, P. (2015a) Assessing cross-sectoral climate change impacts, vulnerability and adaptation: an introduction to the CLIMSAVE project. Climate Change, 128, 153– 167. [doi: 10.1007/s10584-015-1324-3]
- Harrison, P.A., Dunford, R., Savin, C.-M., Rounsevell, M.D.A., Holman, I.P., Kebede, A. & Stuch, B. (2015b) Cross-sectoral impacts of climate change and socio-economic change for multiple European land- and water-based sectors. Climatic Change, 128, 279-292. [DOI 10.1007/s10584-014-1239-4]
- Harrison, P., Dunford, R., Holman, I. & Rounsevell, M. (2016) Climate change impact modelling needs to include cross-sectoral interactions. Nature Climate Change [doi: 10.1038/nclimate3039]
- Harrison, P.A., Dunford, R.W., Holman, I.P., Cojocaru, G., Madsen, M.S., Chen, P.Y., Pedde, S. & Sandars, D. (2018) Differences between low-end and high-end climate change impacts in Europe across multiple sectors. Regional Environmental Change. [DOI: 10.1007/s10113-018-1352-4]
- Henne, P.D., Elkin, C., Colombaroli, D., Samartin, S., Bugmann, H., Heiri, O., Tinner, W. (2013) Impacts of changing climate and land use on vegetation dynamics in a Mediterranean ecosystem: insights from paleoecology and dynamic modelling. Landscape Ecology, 28, 819–833.
- Henne, P.D., Elkin, C., Franke, J., Colombaroli, D., Calò, C., La Mantia, T., Pasta, S., Conedora, M., Dermody, O., Tinner, W. (2015) Reviving extinct Mediterranean forest communities may improve ecosystem potential in a warmer future. Frontiers in Ecology & Environment, 13, 356–362.
- Holman, I., Audsley, E., Berry, P., Brown, C., Bugmann, H., Clarke, E., Cojocaru, G., Dunford, R., Harrison, P.A., Janes, V., Kovats, S., Lafond, V., Li, S., Lobanova, A., Mokrech, M., Rounsevell, M., Sandars, D., Savin, C. & Wimmer, F. (2015) Specification for European model improvement and development. EU FP7 IMPRESSIONS Deliverable D3B.1. Available from <a href="https://www.impressions-project.eu/documents/1">www.impressions-project.eu/documents/1</a>
- Holman, I., Audsley, E., Berry, P., Brown, C., Bugmann, H., Clarke, E., Cojocaru, G., Dunford, R., Fronzek, S., Harrison, P.A., Honda, Y., Janes, V., Kovats, S., Lafond, V., Lobanova, A., Madsen, M.S, Mokrech, M., Nunze, S., Pedde, S., Sandars, D., Savin, C. & Wimmer, F. (2017). Modelling Climate Change Impacts, Adaptation and Vulnerability in Europe. IMPRESSIONS Deliverable D3B.2. Available from <a href="http://www.impressions-project.eu/documents/1">www.impressions-project.eu/documents/1</a>
- Holman, I., Brown C., Carter T., Harrison P. & Rounsevell, M. (2018) Improving the representation of adaptation in climate change impact models. Regional Environmental Change. [doi: 10.1007/s10113-018-1328-4]
- Huang, S., Kumar, R., Flörke, M., Yang, T., Hundecha, Y., Kraft, P., Gao, C., Gelfan, A., Liersch, S., Lobanova, A., et al. (2016) Evaluation of an ensemble of regional hydrological models in 12 largescale river basins worldwide. Climatic Change, 141, 381-397 [doi: 10.1007/s10584-016-1841-8]
- Hölscher, K., Frantzeskaki, N., Holman, I., Pedde, S., Juhasz-Horvath, L., Clarke, E., Schipper, K. & Jäger, J. (2017) Adaptation and mitigation pathways, and synergy mechanisms between them, for the case studies. EU FP7 IMPRESSIONS Project Deliverable D4.2. Available from <u>www.impressions-project.eu/documents/1</u>
- IIASA (2012) SSP database, 2012-2015, International Institute for Applied Systems Analysis. https://tntcat.liasa.Ac.At/SspDb. Accessed 19.9.2016.

- Izaurralde, R., Williams, J., McGill, W., Rosenberg, N. & Jakas, M. (2006) Simulating soil C dynamics with EPIC: Model description and testing against long-term data, Ecological Modellelling, 192, 362–384, [doi:10.1016/j.ecolmodel.2005.07.010]
- Jäger, J., Rounsevell, M., Harrison, P., Omann, I., Dunford, R., Kammerlander, M. & Pataki, G. (2014) Assessing policy robustness of climate change adaptation measures across sectors and scenarios. Climatic Change, 128, 395-407.
- Jäger Jager, J., Rousevell, M.D.A., Harrison, P.A., Omann, I., Dunford, R., Kammerlander, M. and Pataki, G. (2015) Assessing policy robustness of climate change adaptation measures across sectors and scenarios. Climatic Change, 128(3-4): 395-407.
- Jones, B. & O'Neill, B.C., (2016) Spatially explicit global population scenarios consistent with the Shared Socioeconomic Pathways. Environmental Research Letters, 11(8), p.084003. <u>http://iopscience.iop.org/article/10.1088/1748-9326/11/8/084003/meta</u>
- Jones, J., Hoogenboom, G., Porter, C., Boote, K., Batchelor, W., Hunt, L., Wilkens, P., Singh, U., Gijsman, A. & Ritchie, J. (2003) The DSSAT cropping system model. European Journal of Agronomy, 18, 235-265.
- Kebede, A.S., Dunford, R., Mokrech, M., Audsley, E., Harrison, P.A., Holman, I.P., Nicholls, R.J., Rickebusch, S., Rounsevell, M.D.A., Sabaté, S., Sallaba, F., Sanchez, A., Savin, C., Trnka, M., Wimmer, F. (2015) Direct and indirect impacts of climate and socio-economic change in Europe: a sensitivity analysis for key land- and water-based sectors. Climatic Change, 128(3-4): 261-277.
- Kok, K. & Pedde, S. (2016) IMPRESSIONS socio-economic scenarios. EU FP7 IMPRESSIONS Project Deliverable D2.2. Available from <u>www.impressions-project.eu/documents/1</u>
- Kok, K., Christensen, J., Madsen, M., Pedde, S., Gramberger, M., Jäger, J. & Carter, T. (2015) Evaluation of existing climate and socio-economic scenarios. EU FP7 IMPRESSIONS Project Deliverable D2.1. Available from <a href="https://www.impressions-project.eu/documents/1">www.impressions-project.eu/documents/1</a>
- Kok, K., Madsen, M.S., Pedde, S., Christensen, J., Fronsek, S., Nunez Ramos, S., Gramberger, M., Jäger, J., Cojocaru, G., Frantzeskaki, N., Hölscher, K., Tabara, D. & Harrison, P. (2018). Set of integrated high-end, multi-scale scenarios. EU FP7 IMPRESSIONS Project Deliverable D2.4. Available from www.impressions-project.eu/documents/1
- Kotzeva, M., Brandmuller, T, Lupu, I. & Önnerfors, Å. (2016) Urban Europe Statistics on Cities, Towns and Suburbs. European Union. <u>http://ec.europa.eu/eurostat/documents/3217494/7596823/KS-01-16-691-EN-N.pdf/0abf140c-ccc7-4a7f-b236-682effcde10f</u>.
- Li, S., Juhász-Horváth, L., Pedde, S., Pinter, L., Rounsevell, M. & Harrison, P. (2017) Integrated modelling of urban spatial development under uncertain climate futures: A case study in Hungary. Environmental Modelling and Software, 96, 251-264.
- Li, S., Juhasz-Horvath, L., Harrison, P., Pinter, L. & Rounsevell, M. (2016) Mapping population and age structure in Hungary: A residential preference and age dependency approach to disaggregate census data. Journal of Maps, 12(S1), 560-569.
- Liu, J., Williams, J., Zehnder, A. & Yang, H. (2007) GEPIC modelling wheat yield and crop water productivity with high resolution on a global scale. Agricultural Systems, 94, 478–493.

[doi:10.1016/j.agsy.2006.11.019]

- Lobanova, A., Koch, H., Liersch, S. & Hattermann, F., (2016) Impacts of changing climate on the hydrology and hydropower production of the Tagus River Basin. Hydrological Processes [doi:10.1002/hyp.10966]
- Lobanova, A., Liersch, S., Tàbara, J., Koch, H., Hattermann, F. & Krysanova, V. (2017) Harmonizing human-hydrological system under climate change : A scenario-based approach for the case of the headwaters of the Tagus River. Journal of Hydrology 548, 436–447. [doi:10.1016/j.jhydrol.2017.03.015]
- Madsen, M., Maule, C., Christensen, J., Fronzek, S. & Carter, T. (2016) IMPRESSIONS Climate Scenarios. EU FP7 IMPRESSIONS Project Deliverable D2.3.
- Mitchell, T. & Jones, P. (2005) An improved method of constructing a database of monthly climate observations and associated high-resolution grids. International Journal of Climatology, 25, 693 712.
- Murray-Rust, D., Brown, C., van Vliet, J., et al. (2014) Combining agent functional types, capitals and services to model land use dynamics. Environmental Modelling & Software [doi: 10.1016/j.envsoft.2014.05.019]
- Nash, J. & Sutcliffe, J. (1970) River flow forecasting through conceptual models part I—A discussion of principles. J. Hydrol. 0, 282–290.
- Reginster, I. & Rounsevell, M. (2006) Future scenarios of urban land use in Europe. Environment and Planning B: Planning and Design, 33), 619–636. [doi:10.1068/b31079]
- Rickebusch, S. (2010) Maps of land-use change scenario projections for Europe PLUREL project deliverable 1.4.3. http://www.plurel.net/Reports-76.aspx
- Rockel, B. & Woth, K. (2007) Extremes of near-surface wind speed over Europe and their future changes as estimated from an ensemble of RCM simulations. Climatic Change, 81, 267-280.
- Rosenzweig, C., Elliott, J., Deryng, D., Ruane, A., Müller, C., Arneth, A., Boote, K., Folberth, C., Glotter, M., Khabarov, N., Neumann, K., Piontek, F., Pugh, T., Schmid, E., Stehfest, E., Yang, H. & Jones, J. W. (2014) Assessing agricultural risks of climate change in the 21st century in a global gridded crop model intercomparison. PNAS, 111, 3268–73. [doi:10.1073/pnas.1222463110]
- Rounsevell, M., Clarke, E., Dunn, M., Bugmann, H., Cojocaru, G., Cruz, M.J., Harrison, P.A., Horvath, L., Kovats, S., Krysanova, V., Lafond, V., Li, S., Lobanova, A., Lourenço, T.C., Madsen, M.S., Pedde, S., Pinter, L., Snell, R., Savin, C., Tàbara, D. & Terämä, E. (2015) Specification of regional/local scale models and methods. IMPRESSIONS Deliverable D3C.1. Available from <u>www.impressionsproject.eu/documents/1</u>
- Schumacher, S., Bugmann, H., Mladenoff, D. (2004) Improving the formulation of tree growth and succession in a spatially explicit landscape model. Ecological Modelling, 180, 175-194.
- Schumacher, S., Reineking, B., Sibold, J., Bugmann, H. (2006) Modelling the impact of climate and vegetation on fire regimes in mountain landscapes. Landscape Ecology, 21, 539–554.

- Shao G., Bugmann, H., Yan, X. (2001) A comparative analysis of the structure and behaviour of three gap models at sites in northeastern China. Climatic Change, 51, 389–413.
- Smith, B., Prentice, I. & Sykes, M. (2001) Representation of vegetation dynamics in the modelling of terrestrial ecosystems: comparing two contrasting approaches within European climate space, ETEMA Spec. ISSUE Global Ecology & Biogeography, 10, 621–637. [doi:10.1046/j.1466-822X.2001.t01-1-00256.x]
- Taylor, K., Stouffer, R. & Meehl, G. (2012) An overview of CMIP5 and the experiment design. Bulletin of the American Meteorological Society, 93, 485–498. [doi:10.1175/BAMS-D-11-00094.1]
- Temperli, C., Bugmann, H., Elkin, C. (2012) Adaptive management for competing forest goods and services under climate change. Ecological Applications, 22, 2065–2077.
- Terama, E., Clarke, E., Rounsevell, M., Fronzek, S. & Carter, T. (2017) Population structure and urban land use change in an integrated climate modelling framework. Regional Environmental Change. [doi: 10.1007/s10113-017-1194-5]
- Tinch, R., Jäger, J., Omann, I., Harrison, P. A., Wesley, J. & Dunford, R. (2015) Applying a capitals framework to measuring coping and adaptive capacity in integrated assessment models. Climatic Change, 128, 323. [doi: 10.1007/s10584-014-1299-5]
- van Oldenborgh, G., Collins, M., Arblaster, J., Christensen, J.H., Marozke, J., Power, S., Rummukainen, M. & Zhou, T. (2013) Annex I: Atlas of global and regional climate projections. In: Climate Change 2013: The Physical Science Basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change [Stocker, T., D. Qin, G.-K. Plattner, M. Tignor, S. Allen, J. Boschung, A. Nauels, Y. Xia, V. Bex and P. Midgley (eds.)]. Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA, pp. 1311-1393
- Verzano, K. & Menzel, L. (2009) Hydrology in Mountain Regions: Observations, Processes and Dynamics, IAHS-Publication 326, chap Snow conditions in mountains and climate change - a global view, pp. 147–154.
- Weedon, G., Balsamo, G., Belloin, N., Gomes, S., Best, M. & Viterbo, P., (2014) The WFDEI meteorological forcing data set: WATCH Forcing Datamethodology applied to ERA-Interimreanalysis data. Water Resour. Res. 50, 7505–7514. [doi:10.1002/2014WR015638]
- Wittgenstein Centre (2015) Wittgenstein Centre for Demography and Global Human Capital. Wittgenstein Centre Data Explorer Version 1.2. http://www.wittgensteincentre.org/dataexplorer.

## Appendix A

Table A1: Impact and adaptation models used in IMPRESSIONS: their type, sectoral coverage, applications in the project, and the case studies in which they were applied. Modified from Carter et al. (2015).

Model	Туре	Sectoral coverage	Analysis	Case study
AIM/Impact[Health]	STA*	Human health	S, I, C	Glo, Eur, EUx
AIM/Impact[Water]	PRB*	Water resources	С	Glo
M-GAEZ	PRB*	Agriculture	I.	Glo
VISIT	PRB*	Biodiversity	I.	Glo
GLOBIO	IND	Biodiversity	S, I, C	Glo EUx
WaterGAP2	PRB	Water resources	S, I	Glo, Eur, EUx
iPETS	IAM	Multi-sector; population	S	Glo, Eur, EUx
DSK	ABM	Macroeconomy; energy	Р	Glo, Eur
IMAGE	IAM	Agriculture, Land-use	С	Glo
LAGOM	ABM	Production, consumption	S	Eur
CRAFTY 1.0	ABM	Land use	I, C	Eur
IAP2	IAM	Multi-sector	S, P, C	Eur, Sco
rIAM	IAM	Multi-sector	S	Eur
SFARMOD	ECO*	Agriculture	S, I, C	Eur, Sco
ForClim v3.2	PRB	Forest productivity	S, I, C	Eur, Sco
CFFlood Model	PRB*	Coastal and pluvial flooding	S, I	Eur, Sco
WaterGap meta-model (WGMM)	EMU*	Water resources	S, C	Eur, Sco
SPECIES	STA*	Biodiversity	S	Eur, Sco
RUG	CA	Population, urbanisation	S	Eur
Heat-mortality (HEET)	STA	Human health	S	Eur, Hun
SWIM	PRB	Hydrology	S, P, C	Eur, Ibe, Sco, Hun
APORIA	ABM	Agricultural land use	S	Hun
LandClim v1.4	PRB	Forest landscape	S	Ibe
Lyme disease (LYM)	CEL	Human health	S	Sco, Hun
ALLOCATION	DEM	Population, urbanisation	S, C	Hun
SWAT	PRB	Water resources	I.	Ibe

**Type of model:** ABM=agent-based; CA=cellular automata; DEM=demographic; ECO=economic; EMU=processbased model emulator; IAM=integrated assessment; IND=indicator-based; PRB=process-based; STA=statistical; \*=component model of an IAM.

**Type of analysis conducted**: S=scenario runs for case studies; P=adaptation pathway analysis; I= impact response surface (IRS) analysis; C=comparison of models.

**Case study:** Eur=European Union 28 + Norway and Switzerland; EUx=EU external (Central Asia: Kazakhstan, Kyrgyzstan, Tajikistan, Turkmenistan and Uzbekistan); Glo=global; Hun=Hungary; Ibe=Iberia; Sco=Scotland.